# **GRP** – A DIRECT GODUNOV EXTENSION

### MATANIA BEN-ARTZI

ABSTRACT. The classical paper [40] by S. K. Godunov had a revolutionary effect on the field of numerical simulations of compressible fluid flows. Yet, for some twenty years after its publication, it was mostly used by the mathematical community in the Soviet Union. The seminal paper of van Leer [85] has inaugurated the period of universal interest in high-resolution extensions of Godunov's scheme. The first step to take was to modify the (locally) self-similar solution to the Riemann Problem (at discontinuities) by allowing piecewise-polynomial (rather than piecewise-constant) initial data. The GRP (Generalized Riemann Problem) analysis [6] provided analytical solutions (for piecewise-linear data) that could be readily implemented in a high-resolution robust code. This review paper focuses on the evolution of some mathematical aspects of this method. It addresses the three concepts of *consistency*. stability and convergence in the context of compact finite difference (or finite volume) schemes for systems of nonlinear hyperbolic conservation laws. The treatment utilizes the framework of "balance laws", a common viewpoint in relevant physical conservation laws. The first significant observation is that under very mild conditions a weak solution is indeed a solution to the balance law (obtained by a formal application of the Gauss-Green formula). Since highresolution schemes require the computation of several quantities per mesh cell (e.g., slopes), the notion of "consistency" must be extended to this framework. A combination of consistency hypothesis with stability of the scheme leads to a suitable convergence theorem, generalizing the classical convergence theorem of Lax and Wendroff [52]. Finally, the limit functions are shown to be entropy solutions by using a notion of "Godunov compatibility", which serves as a substitute to the entropy condition.

# 1. INTRODUCTION

The seminal paper [40] by S. K. Godunov had a revolutionary effect on the field of numerical simulations of compressible fluid flows. However the initial evolution of this effect was quite slow, when compared to other parallel advances related to the numerical simulations of fluid flows (e.g., vortex methods or finite elements). Eight years after Godunov's publication, the classical book of R.D. Richtmyer and K.W. Morton characterized his paper as follows [77, Section 12.15]:

"In 1959, Godunov described an ingenious method for one-dimensional problems with shocks." Yet, later on in the same section, they add the comment: "The method appears to have been extensively used in the Soviet Union."

We begin this review by a (brief and not comprehensive) account of the developments (spanning the 1960's and 1970's) leading from the Godunov scheme to the

Date: August 31, 2024.

<sup>2020</sup> Mathematics Subject Classification. Primary 65M12; Secondary 35L65, 65M06.

Key words and phrases. Godunov scheme, conservation laws, convergence, discontinuous solutions, Lax-Wendroff theorem, finite volume schemes, high order schemes, Riemann problem, generalized Riemann problem, hyperbolic balance laws, numerical flux.

MUSCL scheme. The latter has inspired an extensive research in high-resolution finite-difference simulations of solutions to nonlinear hyperbolic conservation laws, with emphasis on compressible fluid flows.

Then we concentrate on the GRP (Generalized Riemann Problem) method, which is a direct analytical extension of the Godunov scheme. Our aim is to discuss mathematical aspects of this method. More specifically, we concentrate on three issues:

- Rigorous statement of the **Lipschitz continuity of fluxes**, for any weak solution. This supports an underlying principle of the Godunov scheme, namely, designating the flux as a main object for numerical approximation. On the way, it enables the formulation of the conservation law as a **balance law**.
- Extension of the concept of **consistency**, thus making it applicable to a wide array of schemes, allowing in particular for high order, compact schemes, and taking into account the presence of discontinuities. In particular, our analysis shows that the Godunov scheme is never first-order; it's either of "zero order" or "infinite order", depending on the space of admissible approximations.
- Linking the consistency of the discrete solution to the question of its **convergence** to the solution of the balance law. It involves a renewed consideration of the classical Lax-Wendroff theorem [52].

## 1.1. FROM GODUNOV TO MUSCL.

In subsequent years quite a few finite-difference schemes have been proposed for the numerical simulation of nonlinear hyperbolic conservation laws. They were all (with the exception of the Lax-Wendroff scheme [52]) basically first-order. Ten years after the publication of the book by Richtmyer and Morton, the well-known survey paper by Sod [81] addressed the most prominent and widely used schemes of the time. In order to compare their performances (including computational efficiency) he proposed a test case in the form of a "shock-tube problem". This particular test has been extensively used ever since and, almost fifty years on, is still one of the most popular test cases in current papers in this field.

The results of all the considered schemes showed a great amount of smearing of discontinuities (shocks, contact discontinuities and edges of rarefaction waves). Indeed, in order to get sharper results various "corrective" mechanisms had to be added, such as artificial viscosity, Harten's "artificial compression", a variety of "flux corrections" or hybrid schemes.

Sod's conclusion [81, Section 4] was "Of all the finite difference schemes tested, without the use of corrective procedures, Godunov's and Hyman's methods produced the best results".

Of all the schemes considered, only Godunov's scheme relied on the resolution of the Riemann problem (RP) as its basic tool. In fact its numerical flux is the exact flux as obtained from the Riemann solution. It should be noted that Glimm's scheme ("random choice method" by Chorin's implementation [24]), while relying on the Riemann problem, does not really belong to this category; it is not a finitedifference scheme in the sense that (by randomness) it yields different results when successively applied to the same initial data. It is appropriate at this point to highlight the *fundamental principles embedded* in Godunov's scheme. For years to come they persisted as the "ideological basis" for a substantial part of the numerical simulations of hyperbolic conservation laws, most importantly *finite volume schemes* [35].

## PRINCIPLES OF THE GODUNOV APPROACH

•Determining numerical fluxes by local solution of the exact system.

(1.1)

•The role of upwinding in computing the fluxes.

•Updating the numerical solution via a discretized balance law.

It is fair to say that at this point the whole field was waiting for the next meaningful developments in attempting to obtain a high-order true generalization of the Godunov scheme. This expectation was fulfilled, immediately following Sod's review, in the body of the seminal paper of van Leer [85]. He introduced a secondorder (in space) discretization by allowing piecewise-linear (instead of piecewiseconstant) approximations, linear in each grid cell with jumps (of function and slope) at cell boundaries. Of course such a change implies that the solution (for short time) at each cell-boundary ceases to be the classical self-similar solution to the local RP and leads to a more complex wave structure emanating from the singularity. To account for this extra complexity, van Leer proposed a modification of the solution by representing the deviation from self-similarity by two (curvilinear in space-time) shocks (thus admitting also rarefaction shocks). He then concluded that in this modified setting the values of pressure and velocity near the singularity (constant along each ray in the case of RP) evolve linearly in time. In Example 4.20 below the relevant details are given. In order to avoid oscillations (inherent to second-order schemes) van Leer introduced a "limiter" mechanism, effectively forcing dissipation near strong discontinuities.

Van Leer labeled the resulting scheme as **MUSCL**=" Monotonic Upstreamcentered Scheme for Conservation Laws." The numerical results obtained by this scheme were much superior to all those obtained in Sod's survey [81]. Thus in one stroke the whole field of finite-difference approximations to hyperbolic conservation laws has experienced a definitive jump in terms of accuracy at discontinuities.

We note that in his review [86] van Leer mentions that some of the ideas of MUSCL were already proposed by Kolgan [47]. His paper was not known in the West and in fact, not even by the leading Soviet mathematicians of the time.

As mentioned earlier, in Sod's survey of the group of prominent schemes, only one (Godunov's) relied on the RP solution as a main building block. An immediate consequence of the introduction of MUSCL was the renewed interest in efficient solvers for the RP. A flurry of approaches to "simplified" Riemann solvers ensued: Roe's linearized form [78] as well as a variety of "approximate Riemann solvers" [20], [39, Ch.III.4], [83]. Also consult [82] for an extensive review and references.

The solution of the RP is a hallmark of the Godunov scheme, and consequently also of MUSCL, as well as subsequent developments along this line. This has led to a very fruitful debate, that is still active, some fifty years later. On one hand, from the mathematical aspect, one should invoke any available tool in resolving a given problem; It is no different from the application of PDE techniques in the study of a problem in differential geometry. On the other hand, it should be recognized

that such schemes are primarily designed for use in engineering and physical problems. They should therefore be kept at a reasonable level of simplicity (for coding purposes) as well as being efficient in their use of computing resources.

## 1.2. CONVERGENCE.

In another seminal paper coming from the Soviet Union, Kružkov [49] introduced his "entropy criterion" for uniqueness of weak solutions to (scalar) hyperbolic balance laws. This had strong impact on the study of convergence of finitedifference schemes to the unique entropy solution. Suitable discrete analogs of entropy/entropy-flux pairs were proposed [26, 28, 29, 30, 33, 54, 73, 75, 79]. By the mid 1980's the main goal has been achieved; all monotone (necessarily first-order) "E-schemes", including that of Godunov, were shown to converge to the exact entropy solution, in the case of **scalar** hyperbolic conservation laws [39, Chapter III].

Concerning the question of first-order finite-difference schemes for **systems** of hyperbolic conservation laws (notably isentropic or non-isentropic fluid flows), there have been very few works to date. Of course this is largely due to the absence of theoretical tools (e.g.,total variation or  $L^{\infty}$  estimates), with the remarkable exception of "compensated compactness" (for  $2 \times 2$  systems). We refer to [29, 30], and also to the comprehensive review [35] of the convergence properties of the closely related finite volume methods. Remark that concerning the Godunov scheme for the non-isentropic compressible inviscid flow (in one space dimension), the convergence to a weak solution remains an open problem (and the uniqueness of such a solution is also open).

Turning back to the scalar case, and in light of the computational success of the MUSCL scheme, the (mathematical) research interest turned to convergence properties of the MUSCL (and other second-order) finite-difference schemes. Generally speaking, the discrete solution is shown to satisfy an entropy inequality subject to an appropriate discrete entropy/entropy-flux pair. Also, some structural hypotheses are imposed (e.g., limiting the number of extrema). Then, when the discrete solution converges to a limit (using some compactness property), the limit is indeed an entropy solution. We refer to [12, 17, 22, 41, 42, 45, 54, 65, 68, 74, 87] and references therein for such treatments. It is interesting to recall here the following paragraph from DiPerna's paper [30]: "In the setting of the scalar conservation it remains an open problem to establish convergence of conservative finite difference schemes which are accurate to second order. Stability and convergence results have been obtained so far only for methods which are precisely accurate to first order." Recall that van Leer's original scheme reduces to that of Godunov's when all piecewise-linear approximate solutions are actually piecewise-constant. To the best of the author's knowledge, there is yet no proof for the convergence of the MUSCL scheme (in the scalar case).

## 1.3. MUSCL AND BEYOND.

In the ensuing years a wide panoply of generalizations (or modifications) of the MUSCL scheme (for systems) have been put forward. In all of them the discrete solution is a piecewise-polynomial function, with discontinuities at cell-boundaries.

The primary objective was to enable the numerical resolution of complex (multidimensional) compressible flows. This is still a very active line of research, with a common theme of evaluating the numerical fluxes by some sort of a direct or indirect approximate solution to the local generalized Riemann problem. Some of the most prominent methods are the ADER [84], DG [80], GRP [9], HLL [45], PPM [25], WENO [67, 80]. We mention that there have been various attempts to obtain second-order accuracy while completely circumventing the need for a Riemann solver. They are mathematically interesting but it seems that none of them could lead to a robust scheme for complex flows [42, 44, 50].

The review [86, Section 2.2] deals with high-resolution schemes obtained by "Flux-Corrected Transport" (FCT): a first-order, nonoscillatory scheme is used to obtain an intermediate approximation at the advanced time level; a correction step then removes the large dissipative error made in the first step, obtaining a solution of higher order accuracy. We refer the reader to this review also for an account of the attempts to incorporate the requirement of "Total-Variation-Diminishing" (TVD) into high resolution schemes. Observe that the design of such schemes involves, in addition to the flux evaluations, very elaborate polynomial reconstructions in grid cells.

Henceforth we concentrate on the GRP (Generalized Riemann Problem) methodology, the subject matter of this paper. The paper is organized as follows.

Section 2 introduces the GRP scheme. There have been many versions of the scheme, adapted to engineering and physical applications. A survey (necessarily not exhaustive) of some of them is given, as well as some references to comparative studies involving closely related high-resolution schemes.

Section 3 deals with the basic definition of a hyperbolic "balance law". In order to show that the classically defined weak solutions are indeed solutions to the balance law, the continuity properties of the associated fluxes need to be studied. Theorem 3.2 states that under very general conditions on weak solutions (even those not satisfying any entropy condition) the fluxes are in fact locally Lipschitz continuous.

Section 4 introduces the notion of "approximate fluxes" and their "order of consistency" (Definition 4.15). These notions rely on the order of the spaces used in the approximation (Assumption 4.7). In particular, it is shown (Corollary 4.25) that the order of consistency as introduced here conforms with the previous notion when the latter is applicable. A general definition of *finite volume schemes* (FVS) is introduced (Definition 4.22), that also depends on the order of the approximating space.

It should be emphasized that in case of discontinuous solutions, our definition of consistency may yield different orders than commonly used. Thus, the Godunov scheme (Example 4.18) is of infinite order when applied to piecewise-constant functions but of order zero when applied to piecewise-linear functions. Similarly, the GRP or MUSCL schemes (4.18) are only first-order consistent, while second-order consistency requires smooth solutions.

The foundational "Lax Equivalence Theorem" highlighted the close connection among the three concepts: consistency, stability and convergence. Indeed, in the context of linear evolution equations it asserts that a consistent scheme is stable if and only if it is convergent [77]. Although it was formulated for *linear* evolution equations, it played a decisive role in the development of first order finite-difference

schemes, with special emphasis on hyperbolic conservation laws. The growing use of high-order schemes in this context has made it difficult to adapt the theorem in a straightforward fashion. This is particularly true for nonlinear hyperbolic conservation laws, where the presence (and formation) of discontinuities does not easily allow the examination of "consistency" by standard Taylor expansions. Furthermore, compact (high-order) schemes require the computation of several quantities per mesh cell (e.g., slopes). As we shall see, in this case many of the existing definitions of consistency are not applicable.

Section 5 deals with the convergence of the approximate solutions to the exact solution of the balance law. Recall here DiPerna's observation cited above (Subsection 1.2) stating that "it remains an open problem to establish convergence of conservative finite difference schemes which are accurate to second order. Stability and convergence results have been obtained so far only for methods which are precisely accurate to first order."

Since then, the convergence of various second-order schemes to the unique entropy solution (in the scalar case) has been established as mentioned above (Subsection 1.2). These studies treated specific schemes and employed suitable discrete entropy inequalities. In particular, they had no need to introduce a general concept of consistency as had been done in the first-order case described above.

In Theorem 5.1 the convergence of the approximate solutions to solutions of the balance law is outlined for a general class of high-order finite difference scheme, under certain boundedness conditions. The hypothesis that the scheme is consistent of order q > 0 plays a crucial role. The natural question to be asked is whether or not the limit functions satisfy the entropy condition. In order to provide an affirmative answer the concept of "Godunov compatibility" is introduced in Definition 6.5 and is used in Theorem 6.7. It should be noted that the compatibility condition is based on the assumption that the Godunov scheme converges to an entropy solution; this fact has actually been proved in the scalar case [39] where it is known to be unique and for a class of  $2 \times 2$  systems by DiPerna [30]. We refer also to [29] in the case of isentropic gas dynamics.

## 2. THE GENERALIZED RIEMANN PROBLEM-GRP

The term "generalized Riemann problem" [6, 8] referred originally to the following analytical problem, concerning the initial-value problem for a system of nonlinear hyperbolic conservation laws (in one space dimension):

Consider initial data consisting of linear profiles for  $\pm x > 0$  with a jump at x = 0. Determine the analytical solution for (x, t) in the neighborhood of the singularity at (0, 0).

In the case of constant profiles the problem reduces to the classical Riemann problem, and on the other hand the problem is readily extended to initial data with higher order polynomial profiles (see e.g. [25, 76, 90]).

A variety of solutions have been presented, using techniques such as asymptotic expansions in the original (x, t) coordinates, application of Riemann invariants,

mapping to characteristic coordinates and more. We refer to [9, 13, 18, 19, 41, 53, 59, 71, 84] and references therein.

It should be remarked that the analytical procedure can be viewed as incorporating "physical effects" (in particular entropy variations and other thermodynamical effects) into the solution. In other words, the role of such effects is more prominent in the GRP methodology than in the majority of finite-difference schemes. We refer to [62] for a detailed discussion of this topic. Obviously, this leads to a more complex task of computer implementation of this scheme.

We shall see below (Example 4.21) that the introduction of the **acoustic GRP** provides a good remedy to this difficulty. As a matter of fact it is identical to the second-order ADER scheme.

The GRP analytic solution leads in a straightforward way to an extension of the Godunov scheme, for systems of conservation or "quasi-conservation" laws. The resulting GRP scheme is a robust "shock capturing" method, that has no need for any "adjustable" parameters, apart from enforcing some dissipation via the use of a basic "limiter". This is a mechanism that suppresses oscillatory behavior near sharp discontinuities, a behavior that is typical to any high-order scheme.

As already mentioned, when the discrete profile is piecewise-constant, the GRP scheme reduces precisely to the Godunov scheme. The main highlights of the former conform to the principles of the latter (see (1.1)):

- The numerical flux is the principal object. It is computed at all "singularities", using the analytical solution. These include, naturally, all jumps at "cell boundaries", but optionally also at selected strong shocks, material interfaces and so on.
- Use the flux in a straightforward time-marching of averages of flow variables, via a balance equation.
- Apply once again the GRP analysis to determine slopes at the next time level.

The GRP scheme has been extended and applied in many physical and engineering settings. A partial list includes reacting flows [3, 5], duct flows [7], atmospheric flows on the sphere [11], adaptive meshes [43], boundary value problems [32, 63], multi-fluid multi-dimensional flows [55], shallow water [58], Lagrangian formulation on unstructured meshes [69, 70], flows of real materials [88], relativistic hydrodynamics [91, 92], laminar two-phase two-velocity flow [93]. It was even tested against laboratory experiments [36, 46, 55].

There have been various studies comparing (or combining) the GRP scheme with other well-known schemes. We mention here:

- Comparison between the GRP and gas-kinetic schemes can be found in [61].
- Combination of GRP and DG (Discontinuous Galerkin) can be found in [89].
- Combination of the GRP solution with HLLI Riemann solver can be found in [1].
- Comparison between the GRP and RK4-WENO scheme (1-D and 2-D) can be found in the survey [57].
- Combination of GRP with WENO reconstruction can be found in [31].

## 3. FLUXES-THE HEART OF THE MATTER

We shall now state a very general fact about the Lipschitz continuity of the fluxes for weak solutions of systems of hyperbolic conservation laws. This is in sharp contrast to the fact that the unknowns in the system are in general discontinuous. Clearly a Lipschitz continuous function is more amenable to approximation by various continuous or discontinuous functions (e.g., polynomials or step functions). This accounts for the fundamental role of fluxes in the GRP scheme.

# 3.1. SYSTEMS OF CONSERVATION LAWS IN ONE SPACE DIMENSION.

We consider a system of conservation laws of D unknowns in one space dimension:

(3.1) 
$$u_t + f(u)_x = 0, \quad u, f(u) \in \mathbb{R}^D, \quad (x, t) \in \mathbb{R} \times \mathbb{R}_+,$$

subject to initial data

$$(3.2) u(x,0) = u_0(x), \quad x \in \mathbb{R}.$$

In order to avoid complications caused by the presence of boundaries, we limit our considerations to the pure initial value problem set on the whole line  $\mathbb{R}$ .

Formally, by integration we infer that for every rectangle  $Q = [x_1, x_2] \times [t_1, t_2] \subseteq \mathbb{R} \times \overline{\mathbb{R}_+}$  the following equality holds.

$$(3.3) \quad \int_{x_1}^{x_2} u(x,t_2) dx - \int_{x_1}^{x_2} u(x,t_1) dx = -\Big[\int_{t_1}^{t_2} f(u(x_2,t)) dt - \int_{t_1}^{t_2} f(u(x_1,t)) dt\Big].$$

**Remark 3.1.** Equation (3.3) can be considered as an integrated (formal) form of (3.1), using the Gauss-Green theorem. However, the application of this theorem is certainly not straightforward, since the function u(x,t) is not even continuous (see [37, Section 4.5]). We refer to [23] and [27, Chapter I] for an abstract discussion of this topic. Regarding the right-hand side of (3.3) one needs to keep in mind the following comment concerning the identification of the boundary flux: "the drawback of this, functional analytic, demonstration is that it does not provide any clues on how the  $q_{\mathfrak{D}}$  may be computed from A" [27, Section 1.3].

In fact, the meaning of the x and t derivatives must be clarified since the solutions generate discontinuities, such as shocks or interfaces. As is well known, the concept of a **weak solution** is introduced precisely in order to handle this difficulty [34, Chapter 11], as follows.

For every rectangle  $Q = [x_1, x_2] \times [t_1, t_2] \subseteq \mathbb{R} \times \overline{\mathbb{R}_+}$ , if  $\phi(x, t) \in C_0^{\infty}(Q)$ , then

(3.4) 
$$\int_{t_1}^{t_2} \int_{x_1}^{x_2} [u(x,t)\phi_t + f(u(x,t))\phi_x] dx \, dt = 0.$$

In light of the above comments, the proof of the following theorem is not so obvious [14, Theorem 2.2]:

**Theorem 3.2.** Let u(x,t) be a weak solution to the system (3.1), with initial function  $u_0 \in L^1(\mathbb{R}) \cap L^{\infty}(\mathbb{R})$ .

Assume that u(x,t) satisfies the following properties.

• u(x,t) is locally bounded in  $\mathbb{R} \times \overline{\mathbb{R}_+}$ .

 $x_2$ 

• For every fixed interval  $[x_1, x_2] \subseteq \mathbb{R}$  the mass

(3.5) 
$$m(t) = \int_{x_1}^{\infty} u(x,t) dx$$
 is a well-defined and continuous function of  $t \in \overline{\mathbb{R}_+}$ .

Then we have:

- (i) For every fixed  $[t_1, t_2] \subseteq \mathbb{R}_+$  the integral  $g(x) = \int_{t_1}^{t_2} f(u(x, t)) dt$  is locally Lipschitz continuous in  $x \in \mathbb{R}$ .
- (ii) u(x,t) satisfies the equality (3.3) in every rectangle Q.

**Remark 3.3.** [Alternative continuity hypotheses] We could replace the continuity assumption (3.5) by the stronger assumption that the map  $t \to u(\cdot, t) \in L^{\infty}(\mathbb{R})$  weak<sup>\*</sup> is continuous. This latter assumption is universally imposed when dealing with entropy solutions to nonlinear conservation laws [27, Section 4.5]. However the continuity condition (3.5) is valid for weak solutions that are not necessarily entropy solutions. In fact, it holds for weak solutions that have bounded (locally in time) total variation. This is expressed by Dafermos as "mechanism of regularity transfer from the spatial to the temporal variables" [27, Theorem 4.3.1].

The statement of Theorem 3.2 is closely related to the more fluid dynamical viewpoint: the "conservation law", which is a **partial differential equation**, is replaced by a <u>"balance law"</u>. This latter viewpoint plays a central role in this paper, and is introduced in the following paragraphs.

3.2. **GENERAL SETUP.** We assume the existence of two Banach spaces,  $\mathfrak{U}^{b}$  and  $\mathfrak{U}^{c}$ , with respective norms  $\|\cdot\|_{b}$ ,  $\|\cdot\|_{c}$ , and set

$$\mathfrak{U}=\mathfrak{U}^b\cap\mathfrak{U}^c.$$

**Definition 3.4.** The norm in  $\mathfrak{U}$  is

(3.7) 
$$||w||_{\mathfrak{U}} = ||w||_b + ||w||_c.$$

 $\mathfrak{U}$  is assumed to be a "persistence space" for the class of solutions introduced below, in the sense that they satisfy, for every initial data  $u_0 \in \mathfrak{U}$ ,

(3.8) 
$$t \hookrightarrow u(\cdot, t) \in C(\mathbb{R}_+, \mathfrak{U}^c) \cap L^{\infty}(\overline{\mathbb{R}_+}, \mathfrak{U}^b).$$

Note that as in the case of weak solutions, no uniqueness assumption is imposed at this stage.

We shall also make use of the Fréchet space  $L^1_{loc}(\mathbb{R})$  whose metric is given (for two vector-valued functions f, g) by

(3.9) 
$$d(f,g) = \sum_{N=1}^{\infty} 2^{-N} \frac{\int_{-N}^{N} |f(x) - g(x)| dx}{1 + \int_{-N}^{N} |f(x) - g(x)| dx}, \quad f,g \in L^{1}_{loc}(\mathbb{R}).$$

**Definition 3.5.** Let  $u_0 \in \mathfrak{U}$ . The function  $u(\cdot, t) \in C(\overline{\mathbb{R}_+}, \mathfrak{U}^c) \cap L^{\infty}(\mathbb{R}_+, \mathfrak{U}^b)$  is a solution to the balance law (3.3) corresponding to the partial differential equation (3.1) if the following conditions are satisfied.

• For every  $x \in \mathbb{R}$  and interval  $[t_1, t_2] \subseteq \overline{\mathbb{R}_+}$  the integral  $\int_{t_1}^{t_2} f(u(x, t)) dt$  is well defined, and is a continuous function of  $x \in \mathbb{R}$ .

- For every  $t \ge 0$  and interval  $[x_1, x_2] \subseteq \mathbb{R}$  the integral  $\int_{x_1}^{x_2} u(x, t) dx$  is well defined and is a continuous function of t.
- For every rectangle  $[x_1, x_2] \times [t_1, t_2] \subseteq \mathbb{R} \times \overline{\mathbb{R}_+}$  the balance equation (3.3) is satisfied.

Our definition of a solution to the balance law conforms to that introduced in [27, Chapter I]. In fact, in Dafermos' book the balance equation is assumed to hold for any domain in spacetime. We note that other authors use various other terms, such as the "integral conservation law", and the term "balance law" is applied to a conservation law with a source term.

Definition 3.5 is closely related to the physical interpretation of systems of conservation laws, in particular the Euler system of compressible fluid flow. Furthermore, the balance law serves as the foundation of numerical finite volume schemes; in fact, every interval  $[x_1, x_2]$  is considered as a "control volume" in which the balance law is satisfied between arbitrary time levels  $t_1 < t_2$ .

3.3. THE SCALAR CONSERVATION LAW. We now confine the above discussion to the scalar equation, namely  $u \in \mathbb{R}$ . We refer to the classical paper [49] and to the books [27, 34, 39] for the notion of the Kružkov entropy solution. Note that in Theorem 3.2 we do not need to assume that u(x,t) is an entropy solution.

In this case the function space  $\mathfrak{U}^b$  (see (3.6)) is taken as the space  $L^{\infty}(\mathbb{R})$  while  $\mathfrak{U}^c$  is the space  $L^1(\mathbb{R})$ .

The norm in  $L^p(\mathbb{R})$  is denoted by  $||w||_p$ .

We recall the basic facts concerning this evolution semigroup [39, Chapter 2]:

**Claim 3.6.** The solution semigroup  $S(t) : \mathfrak{U} \hookrightarrow C(\overline{\mathbb{R}_+}, L^1(\mathbb{R}))$  is continuous and satisfies

(i)

$$||S(t)u_0||_{\infty} \le ||u_0||_{\infty}, \quad t \ge 0.$$

(ii)

$$||S(t)u_0 - S(t)v_0||_1 \le ||u_0 - v_0||_1, \quad t \ge 0.$$

Remark in particular that for any fixed interval  $[x_1, x_2] \subseteq \mathbb{R}$  the mass  $m(t) = \int_{x_1}^{x_2} u(x, t) dx$  of the entropy solution u(x, t) is well-defined and, indeed, is a continuous function of  $t \in \overline{\mathbb{R}_+}$ .

## 4. CONSISTENCY-IS GODUNOV'S SCHEME FIRST-ORDER?

Theorem 3.2 implies that a weak solution satisfying certain hypotheses (in particular it need not be an entropy solution) is a solution to the balance law in the sense of Definition 3.5. It is easy to see that conversely, a solution to the balance law is a weak solution of the conservation law (3.1).

For notational simplicity we shall occasionally denote by S(t) the solution operator,

(4.1) 
$$u(\cdot, t) = S(t)u_0(\cdot),$$

even though the solution is not assumed to be unique.

10

To recall the meaning of "consistency", let us consider a vector function u(x,t) taking values in  $\mathbb{R}^D$  and satisfying an evolution equation in  $\mathbb{R} \times \overline{\mathbb{R}_+}$  of the form

$$(4.2) u_t = \Phi(u), \quad t > 0$$

where  $u_t = \frac{\partial}{\partial t}u$  is the partial derivative of u(x,t) with respect to the time variable t, and  $\Phi(u)$  is a general (not necessarily linear) operator involving spatial (with respect to x-variable) derivatives of u (and possibly explicit dependence on x).

Fixing a time interval  $\Delta t > 0$ , the discretization procedure (or "difference method") is aimed at finding a sequence of functions  $\{\widetilde{u^n}(x)\}_{n=0}^{\infty}$ , assumed to approximate the discrete sequence of values of the exact solution  $\{u(x, t_n), t_n = n\Delta t\}_{n=0}^{\infty}$ . The function  $\widetilde{u^0}(x)$  approximates the initial data  $u_0(x)$ . The approximating functions are usually taken from a subspace of the "admissible" functions, those on which the operator  $\Phi$  is acting (in some weak sense).

Generally speaking (for a "one step procedure"), there is a family of operators  $\{\Phi_{\Delta t}, \Delta t > 0\}$  generating the discrete (in time) sequence

(4.3) 
$$\widetilde{u^{n+1}}(x) = \widetilde{u^n}(x) + \Phi_{\Delta t}\widetilde{u^n}(x)$$

The common definition of consistency is the following [77, Section 3.2], [72, Section 5.4], [82, Section 4.1].

**Definition 4.1.** Let q > 0. The discrete scheme (4.3) is consistent of order q with Equation (4.2) in the time interval [0,T] if the exact solution satisfies

(4.4) 
$$u(x, t_{n+1}) - [u(x, t_n) + \Phi_{\Delta t} u(x, t_n)] = O(\Delta t^{1+q}), \quad n\Delta t < T.$$

Observe that the equality (4.4) involves a suitable norm on functions of x, pertinent to the solution and approximating functions.

**Remark 4.2.** In concrete cases, the verification of (4.4) relies on analytical tools, notably Taylor's theorem. This poses a difficulty, since the solution is often discontinuous, as in the case of nonlinear hyperbolic conservation laws. The goal here is to introduce a consistency condition that is meaningful also in the case of discontinuous solutions.

Our definition of consistency is given in Definition 4.15 below. In fact, it involves only **consistency of fluxes.** This is in line with our approach, as already mentioned above, regarding the *balance law* rather than the partial differential equation. Our fundamental Theorem 3.2 establishes the Lipschitz continuity of the fluxes even in the presence of discontinuities, allowing us to define the order of consistency subject to the selected discrete setting. Thus, for example, the Godunov flux is consistent to any order in the setting of piecewise-constant data (Example 4.18), but of *zero order* in the setting of piecewise-linear data (see discussion following Equation (4.16)). In both cases, the uniformity of the grid does not play any role.

We give below a brief (admittedly incomplete) account of the development of the concept of consistency (assuming a uniform grid). In Corollary 4.25 we shall see that the new definition matches the old one under some assumptions (such as uniformity of the grid).

The discrete scheme (4.3) is actually only "semi discrete", since only the (continuous) time is replaced by finite time steps. In practice, in a wide array of schemes

(notably "finite volume") the spatial coordinates are also discretized. Restricting to a one-dimensional framework, a constant mesh size  $\Delta x > 0$  is chosen, so that the ratio

$$\lambda = \frac{\Delta t}{\Delta x}$$
 is a constant.

The approximating function  $\widetilde{u^n}(x)$  is replaced by a discrete sequence  $\widetilde{u^n_{disc}}$  =  $\left\{\widetilde{u_j^n}\right\}_{j=-\infty}^{\infty}$ , that is presumed to approximate the exact values  $\left\{u(x_j, t_n)\right\}_{j=-\infty}^{\infty}$  at the spacetime grid points  $\{x_j = j\Delta x, t_n = n\Delta t\}_{j=-\infty}^{\infty}$ . Accordingly, the semi discrete  $\Phi_{\Delta t}$  is replaced by a fully discrete operator  $\Phi_{\Delta t,\Delta x}$ 

and Equation (4.3) is replaced by a fully (i.e., spatial and temporal) discrete scheme

(4.5) 
$$\widetilde{u_{disc}^{n+1}} = \widetilde{u_{disc}^{n}} + \Phi_{\Delta t, \Delta x} \widetilde{u_{disc}^{n}}.$$

It is clear how to formulate the consistency Definition 4.1 in this case:

**Definition 4.3.** Let q > 0. Denote by  $u_{disc}^n = \{u(x_j, t_n)\}_{j=-\infty}^{\infty}$  the set of values of u at the spatial grid at time  $t_n$ . The discrete scheme (4.5) is consistent of order q with Equation (4.2) in the time interval [0,T] if

(4.6) 
$$u_{disc}^{n+1} - [u_{disc}^n + \Phi_{\Delta t, \Delta x} u_{disc}^n] = O(\Delta t^{1+q}), \quad n\Delta t < T.$$

we now consider the issue of consistency of a fully discrete approximation to the system of conservation laws (3.1).

The classical definition of consistency, introduced by Lax and Wendroff [52], involves a Lipschitz continuous function of 2l variables  $g(\xi_1, \ldots, \xi_{2l}) \in \mathbb{R}^D$ , so that Equation (4.5) can be rewritten as

(4.7) 
$$\widetilde{u_{j}^{n+1}} = \widetilde{u_{j}^{n}} - \lambda \Big[ g(\widetilde{u_{j-l+1}^{n}}, \widetilde{u_{j-l+2}^{n}}, \dots, \widetilde{u_{j+l}^{n}}) - g(\widetilde{u_{j-l}^{n}}, \widetilde{u_{j-l+1}^{n}}, \dots, \widetilde{u_{j+l-1}^{n}}) \Big],$$
$$n = 0, 1, 2 \dots - \infty < i < \infty.$$

**Definition 4.4.** [52, Lax and Wendroff] The scheme (4.7) is consistent with Equation (3.1) if

(4.8) 
$$g(\xi, \dots, \xi) = f(\xi), \quad \xi \in \mathbb{R}.$$

This definition has proved to be very useful in the case of first-order schemes. The Lax-Wendroff theorem [52] ensures that the approximate solutions obtained by a consistent and conservative scheme (4.7), and subject to some boundedness and (weak) convergence hypotheses, actually converge to a weak solution of (3.1). We refer to [28, 33, 38, 48] and references therein for various extensions of this convergence theorem (assuming first-order consistency).

**Remark 4.5.** The methodology of discrete approximations presented here can be viewed as being in either the category of "finite volume schemes" or "finite difference schemes". The concept of "consistency" is fundamental in both. We therefore add here a brief comment concerning the difference between the two approaches. The consistency definition (4.8) applies both to finite difference and finite volume schemes for uniform grids. However, in the case of discrete approximations on non-uniform grids this definition needs to be carefully considered, since a given approximation can be consistent in the "sense of finite differences" but not so in the "sense of finite volume" schemes [35, Remark 21.1].

Practical applications, as well as mathematical interests, require "high order" accuracy, or, using conventional terminology, "high order schemes". Such a requirement can be accommodated by either one of two ways.

- Take a sufficiently large l in (4.7). In other words, extend considerably the "stencil" of dependence when evaluating  $\widetilde{u_j^{n+1}}$ . This in turn involves a more complicated treatment of boundary conditions.
- Instead of considering only "cell averages" (where the value  $u_j^n$  is viewed as an average of the approximate solution in the interval  $(x_j - \frac{\Delta x}{2}, x_j + \frac{\Delta x}{2}))$ , at time  $t_n$ , use more information for each interval, such as slopes or higher moments. This leads to a more complex discrete operator  $\Phi_{\Delta t, \Delta x}$ in Equation (4.5), but enables the use of a "compact scheme", where only the neighboring intervals, centered at  $x_{j\pm 1}$ , are involved in determining the approximate solution  $u_j^{n+1}$ .

The second alternative above is the one that is most widely implemented in various state-of-the-art schemes, such as MUSCL [85], GRP [6, 9, 13], ADER [84], PPM [25], DG [80], WENO [67, 80]. We refer also to the survey paper [82, Section 3.3]. For all these schemes, the concept of consistency must be clearly defined and its connection to the question of convergence to the exact solution should be clarified.

Remark that in our discussion of the system (3.1) the flux function f(u) depends only on the unknown u(x,t). It is very natural (both mathematically and in applications) to try and extend this to more general flux functions. In this case, the issues of consistency and convergence should be addressed. While there is no general framework for such extensions, there are many studies of particular cases, for example [79].

Consider the balance law for systems (3.3). In this section we introduce approximate fluxes associated with it. These fluxes serve in the construction of **compact** schemes, designed to approximate the solution of the balance law.

Let  $k = \Delta t > 0$ . As is common in the literature on finite difference methods (for evolution equations) we set k as the <u>sole</u> parameter in the study. Thus, convergence of approximate solutions to the exact ones will be studied in terms of limits as  $k \to 0$ .

The spatial step is  $h = \Delta x = \lambda^{-1}k$ , where  $\lambda > 0$  is assumed to be fixed.

**Definition 4.6.** The k- spatial grid is the discrete set in  $\mathbb{R}$ ,

$$\Gamma_k = \{x_j = jh\}_{j=-\infty}^{\infty},\,$$

and the grid intervals (or grid cells) are the intervals

$$I_j = \left( x_{j-\frac{1}{2}}, x_{j+\frac{1}{2}} \right), \quad -\infty < j < \infty,$$

where

$$x_{j\pm\frac{1}{2}} = x_j \pm \frac{h}{2}.$$

The spacetime k- grid is the discrete set

(4.9) 
$$\Gamma_k^{spacetime} = \Gamma_k \times \{t_n = nk, \ n = 0, 1, 2, \ldots\}$$

Recall the persistence space  $\mathfrak{U}$  as in Equation (3.8). Given a spatial grid  $\Gamma_k$  we assume that:

**ASSUMPTION 4.7.** (i) There exists a functional subspace  $V^k \subseteq \mathfrak{U}$  having the following property:

The restrictions of the elements of  $V^k$  to any grid interval  $I_j$  constitute a finite dimensional subspace. The dimension of these restrictions is called the **order of**  $V^k$ . Typically, it is a space of piecewise polynomial functions of fixed degree, with possible discontinuities at the boundary points  $\left\{x_{j\pm\frac{1}{2}}\right\}_{j=-\infty}^{j=\infty}$  of every grid interval  $I_j$ .

(ii) There exists a projection P<sup>k</sup> : 𝔄 → V<sup>k</sup>, that does not change averages in grid cells, namely,

(4.10) 
$$\int_{I_j} P^k v(x) dx = \int_{I_j} v(x) dx, \quad -\infty < j < \infty, \ v \in \mathfrak{U}.$$

- **Remark 4.8.** (i) The notation of the space  $V^k$  (and the projection  $P^k$ ) refers explicitly only to the variable parameter k (that determines the spatial step  $h = \lambda^{-1}k$ ). However, this space also depends on our choice of the dimension of its restrictions to grid intervals, such as piecewise-constant ("first order"), piecewise-linear ("second order") and so on.
  - (ii) The operators P<sup>k</sup> are sometimes called "reconstruction operators". They involve suitable interpolations and "slope limiters".

**Notation.** Elements of  $V^k$  will be designated by Greek letters:  $\xi \in V^k$ . There will be <u>no other use</u> of Greek letters throughout the paper (except for the fixed constant ratio  $\lambda = \frac{k}{h}$ ).

4.1. **APPROXIMATE FLUXES.** The approximate fluxes introduced here are intended to be sufficiently general, so as to cover a wide variety of finite volume schemes of any order.

We assume that there exists  $\lambda_0 > 0$  so that for every fixed  $\lambda = \frac{k}{h} < \lambda_0$  and every  $0 < k < \frac{1}{2}T$ ,  $h = \lambda^{-1}k$ , there exists

(4.11) a sequence of continuous functions 
$$\left\{F_{j+\frac{1}{2}}^{\xi}(t), \ 0 \le t < k\right\}_{j=-\infty}^{\infty}$$
, for every  $\xi \in V = V^k$ .

**Definition 4.9 (Approximate Fluxes).** We say that the functions of the family  $\left\{F_{j+\frac{1}{2}}^{\xi}(t), \ 0 \leq t < k\right\}_{j=-\infty}^{\infty}$  are approximate fluxes (in the time interval [0, k)) corresponding to the initial function  $\xi \in V^k$ , if the following finite propagation property is satisfied.

 $F_{j+\frac{1}{2}}^{\xi}(t), \ 0 \le t < k, \ depends \ only \ on \ the \ restriction \ of \ \xi \ to \ I_j \cup I_{j+1}.$ Furthermore, if  $\xi \equiv c = const.$  in  $I_j \cup I_{j+1}$  then  $F_{j+\frac{1}{2}}^{\xi}(t) \equiv f(c).$ 

Note that in this definition the grid points  $\left\{x_{j+\frac{1}{2}}\right\}_{j=-\infty}^{\infty}$  satisfy  $x_{j+\frac{1}{2}} - x_{j-\frac{1}{2}} = h = \lambda^{-1}k$ .

**Remark 4.10.** The assumption above that  $\lambda$  is sufficiently small is the "CFL condition" that enables the finite propagation property of the fluxes.

The terminology of "approximate fluxes" is suggested by the fact that they are viewed as approximating the flux values  $f(u(x_{j+\frac{1}{2}},t))$  at the nodes  $\left\{x_{j+\frac{1}{2}}\right\}_{j=-\infty}^{\infty}$  in a sense that will be made rigorous below (4.12).

**Remark 4.11 (Uniformity of the spatial grid).** While in our treatment the time step k > 0 is constant over the whole mesh, the spatial grid may be non uniform. This is due to the fact that Definition 4.9 relies only on fluxes restricted to cell boundaries. We have chosen to avoid this generality since it leads to notational complications (for example, the underlying discrete spaces  $V^k$  consist of piecewise polynomial functions over cells of variable size).

4.2. CONSISTENCY OF APPROXIMATE FLUXES. We now introduce the key concept of consistency. It is discussed within our framework of balance laws, and therefore it deals exclusively with fluxes. The idea of "consistency" involves a comparison between exact and approximate fluxes, over short time intervals. It should be emphasized that this concept depends in a substantial way on the space of (discontinuous) functions in which the balance law is considered. Preparing the definition we need to introduce suitable families of initial data, contained in the spaces  $V^k$ . For such initial data we assume the short-time existence of unique solutions to the balance law, as in Assumption 4.13 below. We take  $\lambda_0 > 0$  as in Definition 4.9 and consider spacetime grids satisfying  $\lambda = \frac{k}{h} < \lambda_0$ .

**Definition 4.12.** Let  $H \subseteq \bigcup_{0 < k < \frac{1}{2}T} V^k$ . We say that H is an admissible set of

initial data if for every  $\lambda = \frac{k}{h} < \lambda_0$  the set  $H \cap V^k$  is bounded in the  $\mathfrak{U}$  topology and compact in the  $L^1_{loc}(\mathbb{R})$  topology (3.9).

As an example, we can think of H (in the scalar case) as the set of uniformly bounded functions having a finite *total variation*.

**ASSUMPTION 4.13.** Let  $\xi(x) \in V^k$ . Then the balance law (3.3) admits a unique solution in the time interval  $t \in [0, k]$ , subject to the initial condition  $\xi(x)$ . The uniqueness is achieved by imposing suitable constraints, such as "entropy conditions." This solution is denoted henceforth by  $u(x, t; \xi) = S(t) \xi \in \mathfrak{U}, t \in [0, k]$ .

Actually, under some additional boundedness hypotheses, Assumption 4.13 can be verified [64]:

**Claim 4.14.** Let  $V^k \subseteq \mathfrak{U}$  be of any (finite) order. Then for every  $\xi \in V^k$  there exists a unique entropy solution  $u(\cdot, t; \xi) = S(t)\xi(\cdot), t \in [0, k]$ . This solution can be obtained by a constructive procedure, using characteristic curves and generalized Riemann solvers.

**Definition 4.15.** Consider the setup as in Definition 4.9. Let  $q \ge 0$ . The approximate fluxes  $\left\{F_{j+\frac{1}{2}}^{\xi}(t), \ 0 \le t < k\right\}_{j=-\infty}^{\infty}$  are said to be consistent of order q with the balance law (3.3) if for every admissible set of initial data H and all  $\xi \in H \cap V^k$ ,

(4.12) 
$$\left| \int_{0}^{k} \left[ F_{j+\frac{1}{2}}^{\xi}(t) - F_{j-\frac{1}{2}}^{\xi}(t) \right] dt - \int_{0}^{k} \left[ f(u(x_{j+\frac{1}{2}}, t; \xi)) - f(u(x_{j-\frac{1}{2}}, t; \xi)) \right] dt \right|$$
  
$$\leq Ck^{2+q}, \quad -\infty < j < \infty,$$

where C > 0 depends only on H.

**Remark 4.16.** Observe that the order of consistency in Definition 4.15 depends on the choice of the space  $V^k$ . This will be illustrated in Example 4.18 below.

Also, the exponent 2 + q is related to the exponent 1 + q in (4.6). This will be further discussed in Corollary 4.25 below. As already noted in Remark 4.2 the order of consistency may depend on the regularity of the solution. Refer to Subsubsection 4.2.1 below for a detailed analysis of the interplay between regularity and order of consistency.

**Remark 4.17.** Note that the estimate in (4.12) is given by  $Ck^{2+q}$ , where C > 0 depends on H but is independent of j. Typically this dependence is expressed in terms of norms of the restrictions of  $\xi$  to  $I_j$  and neighboring grid intervals.

Of course this can be relaxed by assuming, for instance, that the constant C > 0 is "localized", so that C = C(A), for all j such that  $x_{j\pm\frac{1}{2}} \in [-A, A]$ .

**Example 4.18** (The Godunov Approximate Flux [40]). Let  $V^k$  be of first order, namely, the space of piecewise constant (in grid intervals) functions. Then (if  $\lambda_0$  is sufficiently small by the CFL condition) by definition

(4.13) 
$$F_{j+\frac{1}{2}}^{\xi}(t) = f(u(x_{j+\frac{1}{2}}, t; \xi)), \quad 0 \le t < k, \ -\infty < j < \infty,$$

so that, for this space, the approximate flux is consistent to any order.

Recall that in this case  $u(x_{j+\frac{1}{2}},t;\xi) \equiv \text{const}$  is the solution to the **Riemann problem** subject to the two sided initial data  $\xi_j, \xi_{j+1}$  (constant values in  $I_j, I_{j+1}$ , respectively).

However, as will be seen in Remark 5.4 below, the Godunov approximate flux is only consistent of order zero (q = 0), when used for piecewise-linear functions. In particular, this approximate flux is never first-order!

4.2.1. ORDER OF CONSISTENCY AND REGULARITY. Suppose now that we try to implement the Godunov approximation for the case of second-order spaces (namely,  $V^k$  consists of functions that are linear in grid cells). The approximate flux is therefore

(4.14) 
$$F_{j+\frac{1}{2}}^{\xi}(t) = f(u(x_{j+\frac{1}{2}}, 0+; \xi)),$$

where  $u(x_{j+\frac{1}{2}}, 0+; \xi)$  is the "instantaneous" solution to the Riemann problem subject to the two sided initial data  $\xi_{j+\frac{1}{2}-}, \xi_{j+\frac{1}{2}+}$ , the limiting values of the piecewise linear function  $\xi(x)$  at  $x_{j+\frac{1}{2}}$ .

Let  $u_t(x_{j+\frac{1}{2}}, 0+; \xi)$  be the instantaneous value of the time-derivative of the solution (this is actually the solution to the **Generalized Riemann Problem** (**GRP**) [9, 13]). From

(4.15) 
$$f(u(x_{j+\frac{1}{2}},t;\xi)) = f(u(x_{j+\frac{1}{2}},0+;\xi)) + f'(u(x_{j+\frac{1}{2}},0+;\xi))u_t(x_{j+\frac{1}{2}},0+;\xi)t + \mathcal{O}(t^2).$$

it follows that

$$\begin{split} &\int_{0}^{k}F_{j+\frac{1}{2}}^{\xi}(t)dt - \int_{0}^{k}f(u(x_{j+\frac{1}{2}},t;\xi))dt \\ &= \frac{1}{2}f'(u(x_{j+\frac{1}{2}},0+;\xi))u_{t}(x_{j+\frac{1}{2}},0+;\xi)k^{2} + \mathcal{O}(k^{3}). \end{split}$$

Hence the left-hand side of (4.12) is (4.16)

$$\int_{0}^{k} \left[ F_{j+\frac{1}{2}}^{\xi}(t) - F_{j-\frac{1}{2}}^{\xi}(t) \right] dt - \int_{0}^{k} \left[ f(u(x_{j+\frac{1}{2}}, t; \xi)) - f(u(x_{j+\frac{1}{2}}, t; \xi)) \right] dt$$
  
=  $\frac{1}{2} [f'(u(x_{j+\frac{1}{2}}, 0+; \xi)) u_t(x_{j+\frac{1}{2}}, 0+; \xi) - f'(u(x_{j-\frac{1}{2}}, 0+; \xi)) u_t(x_{j-\frac{1}{2}}, 0+; \xi)] k^2 + \mathcal{O}(k^3).$ 

If no regularity of the solution  $u(x, t; \xi)$  is assumed (in particular, if it is discontinuous) then the approximate flux is only consistent of order zero (q = 0 in (4.12)). However, in regions where the solution is smooth the difference (4.17)

$$f'(u(x_{j+\frac{1}{2}},0+;\xi))u_t(x_{j+\frac{1}{2}},0+;\xi) - f'(u(x_{j-\frac{1}{2}},0+;\xi))u_t(x_{j-\frac{1}{2}},0+;\xi) = \mathcal{O}(k),$$

thus raising the order of consistency to q = 1.

In view of (4.15) the remedy here is to upgrade the approximate flux (4.14) by adding the GRP solution, thus introducing the GRP fluxes.

**Definition 4.19** (**GRP Approximate Flux**). *The GRP approximate flux is given by* 

(4.18) 
$$F_{j+\frac{1}{2}}^{\xi}(t) = f(u(x_{j+\frac{1}{2}}, 0+; \xi)) + f'(u(x_{j+\frac{1}{2}}, 0+; \xi))u_t(x_{j+\frac{1}{2}}, 0+; \xi)t.$$

Now

$$\int_0^k F_{j+\frac{1}{2}}^{\xi}(t)dt - \int_0^k f(u(x_{j+\frac{1}{2}},t;\xi))dt = \mathcal{O}(k^3),$$

so that the <u>order of consistency is q = 1 in all cases</u>. For smooth solutions we obtain second-order consistency (q = 2), since in analogy with (4.18) (4.19)

$$f''(u(x_{j+\frac{1}{2}},0+;\xi))u_t(x_{j+\frac{1}{2}},0+;\xi) - f''(u(x_{j-\frac{1}{2}},0+;\xi))u_t(x_{j-\frac{1}{2}},0+;\xi) = \mathcal{O}(k).$$

Thus, when reduced to the smooth setting, the common statement about the second order consistency of this approximate flux (as well as the MUSCL flux below) is recovered.

**Example 4.20 (MUSCL Approximate Flux).** The derivative  $f'(u(x_{j+\frac{1}{2}}, 0+; \xi))$ in (4.18) depends solely on the Riemann solution. On the other hand, the instantaneous time derivative  $u_t(x_{j+\frac{1}{2}}, 0+; \xi)$  is obtained from the GRP solution. Suppose that we can somehow find an approximation  $v(x_{j+\frac{1}{2}}, 0+; \xi)$  so that

(4.20) 
$$u_t(x_{j+\frac{1}{2}}, 0+; \xi) - v(x_{j+\frac{1}{2}}, 0+; \xi) = \mathcal{O}(k^\beta), \quad \beta \ge 0.$$

Let us define new approximate fluxes by

$$F_{j+\frac{1}{2}}^{\xi}(t) = f(u(x_{j+\frac{1}{2}}, 0+; \xi)) + f'(u(x_{j+\frac{1}{2}}, 0+; \xi))v(x_{j+\frac{1}{2}}, 0+; \xi)t,$$

so that now

$$\int_0^k F_{j+\frac{1}{2}}^{\xi}(t)dt - \int_0^k f(u(x_{j+\frac{1}{2}},t;\xi))dt = \mathcal{O}(k^3) + \mathcal{O}(k^{2+\beta})$$

The MUSCL scheme of van-Leer [85] provides such an approximation with  $\beta = 1$  [9, Appendix D] and we conclude that it is consistent of order q = 1.

**Example 4.21** (Acoustic GRP Approximate Flux). The acoustic GRP flux was introduced in [9, Proposition 5.9] and serves as a particularly simple extension of the Godunov flux. In fact, it also serves as the foundation of the ADER methodology [41, p.807]. It is only applicable if no strong discontinuities are present and in this case it has the same order of consistency as the MUSCL flux [9, Theorem 5.36], namely,  $\beta = 1$  in (4.20). However, in the presence of strong discontinuities it is consistent of order q = 0, hence does not offer a formal improvement of the Godunov flux. It should be noted that in simulations of problems that do not involve strong discontinuities it actually yields much better approximations than those provided by the Godunov scheme [84].

It is now clear how to obtain still higher order of consistency (q = 2 in discontinuous cases): a second-order time derivative is added to the generalized Riemann solution. This has already been implemented in the case of the Euler compressible flow [76].

4.3. CONSISTENCY-COMPARING OLD AND NEW. The approximate fluxes introduced above lead to the construction of approximate solutions by *finite* volume schemes. This construction is introduced here, along with the order of consistency of the ensuing scheme. The compatibility of the new definition of order of consistency with the classical definition is established under suitable assumptions (Corollary 4.25).

In order to conform with the conventional treatment, we consider the general step of the scheme (namely, from  $t_n$  to  $t_{n+1}$ ).

Assumption 4.13 is imposed (see also Claim 4.14), guaranteeing the existence of a unique solution  $u(x,t;\xi), \xi \in V^k$ , to the balance law, in every time step.

In the following definition we assume that  $\left\{F_{j+\frac{1}{2}}^{\xi}(t), 0 \leq t < k\right\}_{i=-\infty}^{\infty}$  are approximate fluxes consistent with the balance law (3.3).

(i) Suppose that there is a map  $S(k): V^k \to \mathfrak{U}$  so that, for Definition 4.22. every  $\xi \in V^k$ ,

(4.21) 
$$\int_{I_j} \widetilde{S(k)} \xi dx - \int_{I_j} \xi dx = -\int_0^k \left[ F_{j+\frac{1}{2}}^{\xi}(t) - F_{j-\frac{1}{2}}^{\xi}(t) \right] dt, \quad -\infty < j < \infty.$$

Then S(k) is called an approximate evolution operator to the balance

(ii) Let  $\left\{ F_{j+\frac{1}{2}}^{\xi}(t), \ 0 \le t < k \right\}_{j=-\infty}^{\infty}$  be approximate fluxes consistent with the balance law (3.3). We say that a family of maps  $\{\Phi^k : V^k \to V^k\}_{k>0}$  is a Finite Volume Scheme (FVS) for the balance law (3.3) if

(4.22) 
$$\Phi^k = P^k \widehat{S(k)}$$

where  $P^k$  is the projection as in (4.10).

The validity of such maps and their connection to the consistency of the fluxes is addressed in Propositions 4.23 and 4.24 below.

Given a time step k > 0, we define a sequence  $\{\theta^n\}_{n=0}^{\infty} \subseteq V^k$  as follows.

First,  $\theta^0 = P^k u_0(x)$ . We construct this sequence successively by letting first

(4.23) 
$$u(x,t-t_n;\theta^n) = S(t-t_n)\theta^n,$$

and then

(4.24) 
$$\theta^{n+1} = P^k(u(x, t_{n+1} - t_n; \theta^n))$$

The set of cell averages of these functions is defined by

$$\left\{\theta_{j}^{n+1} = h^{-1} \int_{I_{j}} \theta^{n+1}(x) dx\right\}_{j=-\infty}^{\infty}, \quad n = 0, 1, 2, \dots$$

For the proof of the following proposition we refer to [14, Proposition 3.18].

**Proposition 4.23.** Assume that the approximate fluxes  $\left\{F_{j+\frac{1}{2}}^{\xi}(t), 0 \leq t < k\right\}_{j=-\infty}^{\infty}$  are consistent of order q, in the sense of Definition 4.15. Then the sequence of cell averages over the intervals  $I_j$  satisfies

(4.25) 
$$\theta_{j}^{n+1} - \theta_{j}^{n} = -\frac{\lambda}{k} \int_{t_{n}}^{t_{n+1}} [F_{j+\frac{1}{2}}^{\theta^{n}}(t-t_{n}) - F_{j-\frac{1}{2}}^{\theta^{n}}(t-t_{n})]dt + \mathcal{O}(k^{1+q}), \\ -\infty < j < \infty.$$

**Proposition 4.24.** Assume that the approximate fluxes  $\left\{F_{j+\frac{1}{2}}^{\xi}(t), 0 \leq t < k\right\}_{j=-\infty}^{\infty}$  are consistent of order q and let  $\Phi^k$  be a FVS as in Definition 4.22. Let  $\theta^n \in V^k$  and

(4.26) 
$$\widetilde{\theta^{n+1}}(x) = \Phi^k(\theta^n) \in V^k.$$

Let

$$\left\{\widetilde{\theta_j^{n+1}} = h^{-1} \int_{I_j} \widetilde{\theta^{n+1}}(x) dx\right\}_{j=-\infty}^{\infty}$$

Then

(4.27) 
$$|\widetilde{\theta_j^{n+1}} - \theta_j^{n+1}| = \mathcal{O}(k^{1+q}), \quad -\infty < j < \infty.$$

where the averages  $\theta_j^{n+1}$  are as in Proposition 4.23.

*Proof.* By (4.21) (and the fact that  $P^k$  does not change averages) we get

(4.28) 
$$\int_{I_j} \Phi^k(\theta^n) dx - \int_{I_j} \theta^n dx = -\int_{t_n}^{t_{n+1}} \left[ F_{j+\frac{1}{2}}^{\theta^n}(t-t_n) - F_{j-\frac{1}{2}}^{\theta^n}(t-t_n) \right] dt,$$
$$-\infty < j < \infty.$$

Thus

$$(4.29) \quad \widetilde{\theta_j^{n+1}} - \theta_j^n = -\frac{\lambda}{k} \int_{t_n}^{t_{n+1}} \left[ F_{j+\frac{1}{2}}^{\theta^n}(t-t_n) - F_{j-\frac{1}{2}}^{\theta^n}(t-t_n) \right] dt, \quad -\infty < j < \infty.$$

Comparing this equality with (4.25) we obtain (4.27).

We can now compare the consistency result of Proposition 4.24 to the classical consistency definition as recalled in the Introduction (Definition 4.3).

Define a discrete time evolution, with  $\Delta t = k$ , by

$$\left[\Phi_{\Delta t,\Delta x}\widetilde{\theta^{n}}\right]_{j} = \widetilde{\theta_{j}^{n+1}} - \widetilde{\theta_{j}^{n}}, \quad -\infty < j < \infty, \quad n = 0, 1, 2, \dots$$

**Corollary 4.25.** Under the assumptions of Proposition 4.24 the discrete operator  $\Phi_{\Delta t,\Delta x}$  is of order q in the sense of Definition 4.3. More explicitly, when viewed as acting on the sequence of averages of the exact solution, it satisfies Equation (4.6).

*Proof.* It is assumed that the discrete operator acts on the exact solution, namely,  $\tilde{\theta}_i^n = \theta_i^n$ . Hence Equation (4.27) can be rewritten as

$$\left|\theta_j^{n+1} - [\theta_j^n + (\Phi_{\Delta t, \Delta x}\theta^n)_j]\right| = \mathcal{O}(k^{1+q}), \quad -\infty < j < \infty.$$

This is therefore identical to Equation (4.6).

Thus our definition, while suitable for discontinuous solutions, is in line with the classical definition, when the latter is applicable.

At this point it should be pointed out that the Godunov scheme, as well as the GRP (and the other high-resolution schemes) are based on one-dimensional analysis. The common practice in two-dimensional settings is to determine fluxes perpendicular to a given side of a grid cell by resorting to the one-dimensional methodology in the perpendicular direction. We refer to [56] for a GRP scheme that uses the local analysis of the tangential component of the momentum. The scheme is then modified by incorporating the transversal effects in the evaluation of the fluxes.

In regular 2-D grids, the conventional approach has been the use of a nonlinear version of the "Strang spatial splitting" [9, Chapter 7]. It has led to remarkably good computational results but to the best of the author's knowledge it is still lacking a substantial analytic treatment. We refer to [10] and references therein.

# 5. CONVERGENCE–THE LAX-WENDROFF THEOREM REVISITED

The question of the convergence of the approximate solution to a solution of the balance law is discussed in this section.

Recall the "Lax-Wendroff Convergence Theorem" [52], in the context of approximations to conservation laws. By using a particular notion of consistency (see Definition 4.4) it yields the convergence of discrete approximate solutions to weak solutions. Introducing a different notion of consistency necessarily forces a revision of this convergence theorem.

Here we address the convergence issue in the framework of "balance laws", in one space dimension.

Our goal is to impose conditions on the FVS (Definition 4.22) that will guarantee the convergence of the approximate solutions to a solution of the balance law (Definition 3.5) at a fixed time t = T as  $k \to 0$ . Observe that since we are dealing with *systems* and do not assume any entropy condition, we cannot infer that such a solution to the balance law is unique.

The consistency result of Proposition 4.24 does not imply such convergence. In fact, it deals with the action, over one time step, of the discrete operator on the *exact solution*. In the construction of the approximate solution at time  $t_{n+1}$ , on the other hand, the operator acts on the *approximate solution* obtained at time  $t_n$ . It is given by (see Definition 4.22)

(5.1) 
$$\widehat{\theta^{n+1}}(x) = \Phi^k(\widetilde{\theta^n}) \in V^k, \quad n = 0, 1, 2, \dots$$

Thus, the procedure produces errors that accrue at each time step and do not necessarily vanish at the final time t = T as the time step is refined.

The above discussion can simply be summarized by saying that once *consistency* is assumed then *stability* is (in one form or another) needed in order to ensure convergence. In the *linear* case, this is precisely the claim of the celebrated "Lax equivalence theorem" [77, Section 3.5].

At this stage, it is useful to recall the two main approaches to convergence.

- "compactness"—establishing the boundedness of the discrete solutions in a stronger space that is compactly embedded in the expected convergence space. In the case of discontinuous solutions this is universally carried out in total variation spaces.
- "stability"-imposing some boundedness assumptions on the discrete solutions and using consistency in order to control the accumulation of errors.

The second approach is what can be referred to as the "Lax-Wendroff methodology". It necessarily assumes the existence of an exact solution but, on the other hand, the assumptions imposed on the discrete solutions are typically easier to verify in a concrete computation.

Our presentation here is in the framework of the second approach.

We remark that in the case of a linear evolution equation (even in Banach space) stability (with a suitable assumption on the action of the discrete operator on the residual terms) is sufficient to establish convergence [28]. See also Remark 4.5. However there is no similar result that is applicable to the case of interest here, namely, nonlinear hyperbolic balance laws.

The special consistency condition (4.8), used in establishing the Lax-Wendroff convergence theorem, must be replaced by the notion of consistency as in Definition 4.15.

We shall impose certain conditions that will guarantee that the approximate solution constructed in (5.1) converges to a solution of the balance law.

5.1. THE CONVERGENCE THEOREM. Fix T > 0. Recall the construction of the discrete (in time) sequence of short time exact solutions (see Claim 4.14 and (4.24))

(5.2) 
$$\theta^{n+1}(x) = P^k(u(x, t_{n+1} - t_n; \theta^n)) \in V^k,$$
$$n = 0, 1, 2, \dots, N-1, \ N = N(k) = k^{-1}T,$$

and the sequence of approximate solutions (5.1)

(5.3) 
$$\widetilde{\theta^{n+1}}(x) = \Phi^k(\widetilde{\theta^n}) \in V^k, \quad n = 0, 1, 2, \dots, N-1.$$

For both sequences the initial data is given by taking the projection of the initial function  $u_0 \in \mathfrak{U}$  on the subspace  $V^k$ 

(5.4) 
$$\theta^0 = \widetilde{\theta^0} = P^k u_0 \in V^k.$$

We assume that the conditions of Definition 4.9 (and in particular the CFL condition) are satisfied, so that approximate fluxes can be constructed. We shall further assume that these fluxes are consistent of order q > 0 (Definition 4.15). Since the projection  $P^k$  does not change averages in cells, it follows that for all grid

intervals  $I_j$ ,

(5.5) 
$$\int_{I_j} \theta^{n+1}(x) dx - \int_{I_j} \theta^n(x) dx \\ = -\int_{t_n}^{t_{n+1}} \left[ f(u(x_{j+\frac{1}{2}}, t - t_n; \theta^n)) - f(u(x_{j-\frac{1}{2}}, t - t_n; \theta^n)) \right] dt, \\ -\infty < j < \infty,$$

and from (4.21) that for all grid intervals  $I_j$ ,

(5.6) 
$$\int_{I_j} [\widetilde{\theta^{n+1}}(x) - \widetilde{\theta^n}(x)] dx$$
$$= -\int_{t_n}^{t_{n+1}} [F_{j+\frac{1}{2}}^{\widetilde{\theta^n}}(t-t_n) - F_{j-\frac{1}{2}}^{\widetilde{\theta^n}}(t-t_n)] dt, \ -\infty < j < \infty.$$

We construct a function  $\widetilde{\Upsilon^k}(x,t)$  as follows.

(5.7) 
$$\widetilde{\Upsilon^{k}}(x,t) = \frac{1}{k} [(t_{n+1} - t)\widetilde{\theta^{n}}(x) + (t - t_{n})\widetilde{\theta^{n+1}}(x)], \quad t \in [t_{n}, t_{n+1}], \\ n = 0, 1, \dots, N-1.$$

Observe that  $t_n = nk$  depends on k.

Instead of the classical Lax-Wendroff theorem we get here the following theorem. For the (rather long) proof see [14, Theorem 4.1].

**Theorem 5.1.** Assume that the FVS (5.3) is consistent of order q > 0. Let  $\{k_m \downarrow 0\}$  be a decreasing sequence of time steps. Let  $u_0 \in \mathfrak{U}$  (see (3.6)) and let  $\left\{\widetilde{\Upsilon^{k_m}}(x,t)\right\}_{m=1}^{\infty}$  be the corresponding functions defined in (5.7). Suppose that

- - (i) The sequence  $\left\{\widetilde{\Upsilon^{k_m}}(x,t)\right\}_{m=1}^{\infty}$  is uniformly bounded in  $L^{\infty}([0,T], L^{\infty}(\mathbb{R}))$ . (ii) The sequence  $\left\{\widetilde{\Upsilon^{k_m}}(x,t)\right\}_{m=1}^{\infty}$  converges in  $C([0,T], L^1_{loc}(\mathbb{R}))$  to a function v(x,t) (in particular it is uniformly bounded in this space).

Then v(x,t) is a solution of the balance law (3.3) in  $\mathbb{R} \times [0,T]$ .

Remark 5.2. The boundedness and convergence hypotheses in the theorem can be formulated in terms of the discrete solutions  $\tilde{\theta}^n(x)$  as follows, where  $N_m = k_m^{-1}T$ .

- The set  $\left\{\left\{\widetilde{\theta^{n}}(x)\right\}_{n=1}^{N_{m}} \subseteq V^{k_{m}}\right\}_{m=1}^{\infty}$  is uniformly bounded in  $\mathfrak{U}$  (in the topology (3.7)).
- There exists a function  $v(\cdot, t) \in C([0, T], L^1_{loc}(\mathbb{R}))$  so that

(5.8) 
$$\lim_{m \to \infty} \sup_{1 \le n \le N_m} d(\widetilde{\theta^n}(x), v(x, t_n)) = 0$$

where the metric d(y, z) is given in (3.9).

It was shown (Example 4.18) that the Godunov approximate fluxes on piecewiseconstant functions are consistent of any order while the GRP upgrading (Definition 4.19) is consistent of (at least) first order, hence the following corollary holds.

22

- Corollary 5.3 (Godunov, GRP and MUSCL convergence). (i) Let the FVS (5.3) be given by the approximate Godunov fluxes (Example 4.18), confined to piecewise-constant functions. Then the limit of any convergent sequence, subject to the hypotheses (i)-(ii) of Theorem 5.1, is a solution to the balance law.
  - (ii) Let the FVS (5.3) be given by the approximate GRP fluxes (4.18), where the space V<sup>k</sup> of approximating functions can be of any finite order. Then the limit of any convergent sequence, subject to the hypotheses (i)-(ii) of Theorem 5.1, is a solution to the balance law.
  - (iii) Let the FVS (5.3) be given by the approximate MUSCL fluxes (Example 4.20), where the space  $V^k$  of approximating functions can be of any finite order. Then the limit of any convergent sequence, subject to the hypotheses (i)-(ii) of Theorem 5.1, is a solution to the balance law.

**Remark 5.4.** A fundamental assumption in Theorem 5.1 is that the approximate fluxes are consistent of order q > 0. Recalling the discussion in Subsection 4.2.1 it follows that when the Godunov approximate fluxes are implemented for piecewiseconstant functions, any limit function is a solution to the balance law, as stipulated by Corollary 5.3. On the other hand, taking a (spatially) second-order approximation (namely, piecewise-linear functions), and still using the Godunov approximate fluxes (4.14), the order of consistency is q = 0. The convergence theorem is not applicable and convergence may fail. On the other hand, as is stated in Corollary 5.3, implementing the GRP or MUSCL fluxes raises the order to q = 1 and ensures that any limit function is a solution of the balance law.

These considerations are convincingly demonstrated in the numerical examples worked out in [62], where the aforementioned two possibilities for approximate fluxes (with piecewise-linear data) were tested (see Figs. 6.2 and 6.5 there).

5.2. A MEASURE THEORY LEMMA. Throughout the rest of this section we fix a T > 0. The time steps to be considered will be of size  $k = \frac{1}{N}T$  for an integer N > 1.

Our final goal (Corollary 5.8) is to prove that the *grid averages* of the approximate solutions converge to the solution of the balance law obtained in Theorem 5.1. In proving this, some basic measure-theoretic facts are established.

**Definition 5.5.** Given the spacetime grid  $\Gamma_k^{spacetime}$  (4.9) and a function  $Y(x,t) \in L^1_{loc}(\mathbb{R} \times [0,T])$ , we denote by  $Y^{k,av}(x,t)$  the space-averaged function that consists of the averages in grid intervals,

(5.9) 
$$Y^{k,av}(x,t) = \frac{1}{h} \int_{I_j} Y(z,t) dz, \quad (x,t) \in I_j \times [0,T],$$
$$I_j = (x_{j-\frac{1}{2}}, x_{j+\frac{1}{2}}), \quad -\infty < j < \infty.$$

We have

$$\int_{I_j} Y^{k,av}(x,t) dx = \int_{I_j} Y(x,t) dx$$

and

$$\int\limits_{I_j}Y^{k,av}(x,t)dx\Big|=\int\limits_{I_j}|Y^{k,av}(x,t)|dx$$

So integrating over any  $[a, b] \subseteq [0, T]$  yields

(5.10) 
$$\int_{I_j \times [a,b]} |Y^{k,av}(x,t)| dx dt \leq \int_{I_j \times [a,b]} |Y(x,t)| dx dt.$$

It follows that if the bounded set  $K \subseteq \mathbb{R} \times [0,T]$  is a union of such rectangles then

(5.11) 
$$\int_{K} |Y^{k,av}(x,t)| dx dt \leq \int_{K} |Y(x,t)| dx dt.$$

Using a density argument we now obtain:

**Claim 5.6.** Let  $Y(x,t) \in L^1_{loc}(\mathbb{R} \times [0,T])$ , then for every bounded  $K \subseteq \mathbb{R} \times [0,T]$ ,

(5.12) 
$$\lim_{k \to 0} \int_{K} |Y^{k,av}(x,t) - Y(x,t)| dx dt = 0.$$

Claim 5.6 entails the following lemma.

**Lemma 5.7.** Let  $\{w_m(x,t)\}_{m=1}^{\infty} \subseteq C([0,T], L^1_{loc}(\mathbb{R}))$  be a sequence of functions that converges to a function w(x,t) in the sense that

$$\lim_{m \to \infty} w_m(\cdot, t) = w(\cdot, t), \quad in \ C([0, T], L^1_{loc}(\mathbb{R})).$$

Let  $\{k_m \downarrow 0\}$  be a decreasing sequence. Then the sequence of the corresponding average functions  $\{w_m^{k_m,av}(x,t)\}_{m=1}^{\infty}$  converges to w(x,t) in  $L^1_{loc}(\mathbb{R}\times[0,T])$ .

*Proof.* Let  $K \subseteq \mathbb{R} \times [0,T]$  be bounded. Then in view of (5.11)

$$\lim_{m \to \infty} \int_{K} |w_m^{k_m, av}(x, t) - w^{k_m, av}(x, t)| dx dt = 0,$$

and in view of (5.12)

$$\lim_{m \to \infty} \int_{K} |w^{k_m, av}(x, t) - w(x, t)| dx dt = 0.$$

**Corollary 5.8.** Assume the conditions of Theorem 5.1 and define the sequence of piecewise-constant functions

(5.13) 
$$\widetilde{\Upsilon^{k_m,av}}(x,t) = \frac{1}{h} \int_{I_j} \widetilde{\Upsilon^{k_m}}(z,t) dz, \quad (x,t) \in I_j \times [0,T],$$
$$I_j = (x_{j-\frac{1}{2}}, x_{j+\frac{1}{2}}), \quad -\infty < j < \infty.$$

Then the sequence  $\left\{\Upsilon_m^{k_m,av}(x,t)\right\}_{m=1}^{\infty}$  converges to v(x,t) in  $L^1_{loc}(\mathbb{R}\times[0,T])$ .

This corollary is of great practical significance, as it states that the solution to the balance law can be recovered from the cell averages of the approximate solutions. Obviously, these averages are easier to obtain, in the computational procedure, than the full (piecewise-polynomial) approximate solutions.

## 6. GODUNOV COMPATIBILITY AND ENTROPY

A major difficulty in the theory of nonlinear balance laws is that the limiting solutions obtained in Theorem 5.1 need not be unique. Recall that a solution to the balance law (Definition 3.5) is necessarily a weak solution to the corresponding conservation law (3.1). The entropy condition (see Subsection 3.3 above), essentially the only tool available for establishing uniqueness, has been applicable only in the scalar case [27, 39] (and some  $2 \times 2$  systems [66]).

It is well-known that the Godunov FVS  $\Phi^{k,G}$  provides a "reference frame" to full classes of (first order, scalar) approximate solutions, for example to all "*E*-schemes" [39, Chapter 3, Lemma 4.1]. Also, for a class of 2 × 2 systems it was shown in [30] that all limits of approximate solutions obtained by the Godunov scheme are entropy solutions. These observations seem to justify the introduction of the concept of "Godunov-compatible" schemes (Definition 6.5). It is based on the Assumption 6.4 that the Godunov FVS converges to a unique entropy solution. it is then shown that the approximate solutions produced by Godunov-compatible schemes converge to the same solution.

The treatment here may be compared to that of the Glimm scheme: Under suitable conditions all weak solutions obtained as limits satisfy the entropy condition [51, Theorem 2.2] and the solution is unique in the class of approximate solutions obtained by the front tracking method [21] (see also [27, Chapter XIV]).

Let us first consider the Godunov FVS as introduced in Example 4.18 (for general systems, not only scalar). The scheme is used for first order (namely, piecewise constant) spaces  $V^k$ . The notation  $\Phi^{k,G}$  is used for the Godunov FVS.

In this case the FVS yields a discrete sequence  $\left\{\widetilde{\theta^{n,G}}(x)\right\}_{n=0}^{N}$  of piecewise constant functions as in (5.1). Thus

(6.1) 
$$\widetilde{\theta^{n,G}}(x) = \widetilde{\theta_j^{n,G}}, \ x \in I_j, \ -\infty < j < \infty.$$

The values  $\left\{\widetilde{\theta_j^{n,G}}\right\}_{j=-\infty}^{\infty}$  satisfy

(6.2)  

$$\begin{aligned}
\widetilde{\theta_j^{n+1,G}} &- \widetilde{\theta_j^{n,G}} \\
&= -\lambda \Big[ f(u^G(x_{j+\frac{1}{2}}, t-t_n; \widetilde{\theta^{n,G}})) - f(u^G(x_{j-\frac{1}{2}}, t-t_n; \widetilde{\theta^{n,G}})) \Big], -\infty < j < \infty,
\end{aligned}$$

where  $u^G(x_{j+\frac{1}{2}}, t - t_n; \widetilde{\theta^n}) \equiv const$  is the solution to the Riemann problem at  $x = x_{j+\frac{1}{2}}$ .

As in Equation (5.7) we can now use the set  $\left\{\widetilde{\theta^{n,G}}(x)\right\}_{n=0}^{N}$  in order to define the function  $\widetilde{\Upsilon^{k,G}}(x,t)$  for the Godunov scheme.

6.1. THE SCALAR GODUNOV SCHEME. It is well-known that in the scalar case the Godunov scheme possesses the same boundedness and contraction properties as the exact solution to the balance law (Claim 3.6):

**Claim 6.1** ([39, Section 3.3]). The sequence of solutions to the Godunov scheme satisfies

$$\begin{split} \|\widetilde{\theta^{n+1,G}}\|_{\infty} &\leq \|\widetilde{\theta^{n,G}}\|_{\infty}, \quad n = 0, 1, 2, \dots, N-1. \\ \bullet \ If \ \widetilde{\theta^{0,G}} \ and \ \widetilde{\chi^{0,G}} \ are \ two \ piecewise \ constant \ functions \ and \ \left\{\widetilde{\theta^{n,G}}(x)\right\}_{n=0}^{N}, \\ \left\{\widetilde{\chi^{n,G}}(x)\right\}_{n=0}^{N} \ are \ the \ corresponding \ solutions \ by \ the \ Godunov \ scheme, \ then \\ \|\widetilde{\theta^{n+1,G}} - \widetilde{\chi^{n+1,G}}\|_{1} \leq \|\widetilde{\theta^{n,G}} - \widetilde{\chi^{n,G}}\|_{1}, \quad n = 0, 1, 2, \dots, N-1. \end{split}$$

The fact that the approximate solutions derived by the Godunov scheme converge to the unique entropy solution is a fundamental fact of the theory of discretization of (scalar) conservation laws:

**Claim 6.2** ([39, Chapter 3, Theorem 4.1]). Assume that the CFL condition is satisfied and also that the initial function  $u_0$  has finite total variation. Then the limit

$$v(x,t) = \lim_{k \to 0} \widetilde{\Upsilon^{k,G}}(x,t)$$

exists in  $C([0,T], L^1_{loc}(\mathbb{R}))$  and is the unique entropy solution of the balance law.

**Remark 6.3.** Note that the condition that  $u_0$  has finite total variation is not really needed [35, Theorem 29.2].

6.2. THE CASE OF SYSTEMS. Recall (5.9)) that the projection of  $\xi \in V^k$  on the space of piecewise constant functions, namely, the set of averages in the grid intervals, is designated as

(6.3) 
$$\xi^{k,av}(x) = h^{-1} \int_{I_j}^{r} \xi(z) dz, \quad x \in I_j = (x_{j-\frac{1}{2}}, x_{j+\frac{1}{2}}), \quad -\infty < j < \infty,$$

and satisfies (compare (5.10))

(6.4) 
$$\int_{\mathbb{R}} |\xi^{k,av}(x)| dx \leq \int_{\mathbb{R}} |\xi(x)| dx.$$

For general systems consider the Godunov scheme and recall (Corollary 5.3) that under suitable hypotheses the approximate solutions converge to a solution of the balance law. We now impose the following **fundamental hypothesis on the Godunov scheme** regarding the uniqueness of these solutions.

**ASSUMPTION 6.4.** [Godunov Scheme] Let  $\mathfrak{B}_K$  be the ball of radius K > 0 in  $\mathfrak{U}$  (see (3.6)) and let  $u_0 \in \mathfrak{B}_K$ . Let  $\theta^{0,G} = u_0^{k,av}$ . The Godunov scheme  $\Phi^{k,G}$ , applied to  $\theta^{0,G} = u_0^{k,av}$  converges to a unique solution of the balance law. More precisely, if  $\Phi^{k,G}$  is the FVS in Theorem 5.1 then, under the hypotheses of the theorem, all limits of subsequences  $\widehat{\Upsilon^{k_m,G}}(x,t)$  obtained in the theorem are identical.

Furthermore, if  $v_0 \in \mathfrak{B}_K$  is another initial function and  $\psi^{0,G} = v_0^{k,av}$ , then

(6.5) 
$$\|\Phi^{k,G}\theta^{0,G} - \Phi^{k,G}\psi^{0,G}\|_1 \le (1+Ck)\|\theta^{0,G} - \psi^{0,G}\|_1,$$

where C > 0 depends only on K.

Consider a space  $V^k$  (of any order) and an FVS as in (5.3). In view of (4.10) and (4.21) the map  $\Phi^k$  is conservative:

(6.6) 
$$\int_{\mathbb{R}} \Phi^k \xi(x) dx = \int_{\mathbb{R}} \xi(x) dx, \quad \xi \in V^k.$$

26

•

Let  $\left\{\widetilde{\theta^n} \in V^k\right\}_{n=0}^N$  be the discrete set of approximate solutions, as constructed in (5.3). From them we obtain the function  $\widetilde{\Upsilon^k}(x,t)$  defined in (5.7) and the "average function"  $\widetilde{\Upsilon^{k,av}}(x,t)$  as in (5.9). The set of cell averages of  $\widetilde{\theta^n}$  is denoted by  $\left\{\widetilde{\theta_j^n} = h^{-1} \int_{I_j} \widetilde{\theta^n}(x) dx\right\}_{j=-\infty}^\infty$ .

The following definition encapsulates the meaning of the FVS  $\Phi^k$  as being compatible with the Godunov scheme.

**Definition 6.5.** [Godunov Compatibility] The FVS  $\Phi^k$  (consistent of order q > 0) is compatible with the Godunov scheme if the following conditions hold.

(i) The FVS  $\Phi^k$  coincides with the Godunov scheme on piecewise constant functions; if  $\xi \in V^k$  is piecewise constant then

(6.7) 
$$\Phi^k \xi = \Phi^{k,G} \xi$$

(ii) Let H be an admissible set (Definition 4.12). Then

(6.8) 
$$\int_{\mathbb{R}} |\Phi^k \xi(x) - \Phi^{k,G} \xi^{k,av}(x)| dx = o(k), \quad \xi \in H,$$

where o(k) is uniform for all  $\xi \in H$ .

**Remark 6.6.** [*Explaining* (6.8)] This remark is intended as a motivation, certainly not a proof, for (6.8).

Let  $\eta(x) = \Phi^k \xi(x)$  and  $\chi(x) = \Phi^{k,G} \xi^{k,av}(x)$ . Observe that in light of (4.12) and (4.25), for a fixed index  $j \in \mathbb{Z}$ ,

(6.9)  

$$\int_{I_{j}} [\eta(x) - \xi(x)] dx = -\int_{0}^{k} [F_{j+\frac{1}{2}}^{\xi} - F_{j-\frac{1}{2}}^{\xi}] dt$$

$$= -\int_{0}^{k} \left[ f(u(x_{j+\frac{1}{2}}, t; \xi)) - f(u(x_{j-\frac{1}{2}}, t; \xi)) \right] dt + \mathcal{O}(k^{2+q})$$

$$= \int_{I_{j}} [u(x, k; \xi)(x) - \xi(x)] dx + \mathcal{O}(k^{2+q}).$$

(See Assumption 4.13 for the definition of  $u(x, t; \xi)$ ).

Since the Godunov scheme yields the exact mean value,

(6.10) 
$$\int_{I_j} [\chi(x) - \xi^{k,av}(x)] dx = \int_{I_j} [u(x,k;\xi^{k,av})(x) - \xi(x)] dx.$$

Subtracting (6.10) from (6.9) yields

(6.11) 
$$\int_{I_j} (\eta(x) - \chi(x)) dx = \int_{I_j} [u(x,k;\xi)(x) - u(x,k;\xi^{k,av})(x)] dx + \mathcal{O}(k^{2+q}).$$

Assuming that the exact solution u is a contraction in  $L^1$ , in conjunction with the finite propagation speed property leads to

(6.12) 
$$\int_{I_j} |\eta(x) - \chi(x)| dx \le C_1 \sum_{m=j-l}^{j+l} \int_{I_m} |\xi(x) - \xi^{k,av}(x)| dx + \mathcal{O}(k^{2+q}),$$

where  $C_1 > 0$  and the integer  $l \ge 1$  are independent of j.

If the projection  $P^k$  (see (4.10)) involves a slope limiter, then the right-hand side of (6.12) is estimated by  $C_2k^2(1 + \mathcal{O}(k^q))$ , and summation over j leads to

(6.13) 
$$\int_{\mathbb{R}} |\eta(x) - \xi(x)| dx \le C_2 k (1 + \mathcal{O}(k^q)).$$

Thus, (6.8) can be understood as assuming that in at most  $o(k^{-1})$  grid cells there is a significant discrepancy between  $\Phi^k \xi$  and the piecewise-constant function  $\Phi^{k,G} \xi^{k,av}$ obtained by application of the Godunov scheme.

For the proof of the following theorem we refer to [14, Theorem 5.7].

**Theorem 6.7.** Assume the validity of Assumption 6.4 and that the FVS  $\Phi^k$  is consistent of order q > 0 and compatible with the Godunov scheme. Let  $\left\{ \widetilde{\theta^n} \in V^k \right\}_{n=0}^N$   $(N = k^{-1}T)$  be the discrete set of approximate solutions, as constructed in (5.3). Then the limit function obtained in Theorem 5.1 is unique, namely, under the hypotheses of the theorem there is a unique limit function for all converging subsequences.

**Remark 6.8.** [Godunov Scheme and Entropy] Systems that allow for entropy/ entropy-flux formulations play a special role in the study of balance laws. This is true in particular in various (hyperbolic) models of fluid dynamics. In such cases, Assumption 6.4 can be relaxed, requiring only that all possible limit functions are entropy solutions. As is well-known, this requirement is not sufficient to ensure uniqueness. However, in this case Theorem 6.7 can be modified (under the same hypotheses) to state that all possible limits obtained by the FVS  $\Phi^k$  are entropy solutions.

**Example 6.9.** [Isentropic Gas Dynamics] Consider the Euler system of compressible, isentropic flow in one space dimension:

(6.14) 
$$\rho_t + (\rho u)_x = 0,$$
$$(\rho u)_t + (\rho u^2 + p(\rho))_x = 0,$$

subject to initial conditions

$$\rho(x,0) = \rho_0(x) \ge 0, \quad u(x,0) = u_0(x), \quad x \in \mathbb{R}$$

Here  $\rho$  is the density, u is the velocity and the gas is polytropic:  $p = k\rho^{\gamma}$  with  $1 < \gamma \leq \frac{5}{3}$ . Then we have the following corollary to Theorem 6.7.

**Corollary 6.10.** Suppose that the FVS  $\Phi^k$  is consistent of order q > 0 and compatible with the Godunov scheme. Let  $\left\{\widetilde{\theta^n} \in V^k\right\}_{n=0}^N$   $(N = k^{-1}T)$  be the discrete set of approximate solutions, obtained by applying  $\Phi^k$  to the system (6.14). Then all limit functions obtained in Theorem 5.1 are entropy solutions of the system.

*Proof.* It is shown in [29] that, under some additional conditions on the initial data, the approximate solutions obtained by the Godunov scheme converge to entropy solutions of the system.  $\Box$ 

## 7. SUMMARY

A main ingredient in the Godunov legacy is the construction of the <u>numerical flux</u> by invoking a local analytic solution to the nonlinear hyperbolic conservation law, subject to discretized initial data. The fact that this flux is Lipschitz continuous (Theorem 3.2) implies that it is the proper object to be approximated, even in the presence of various discontinuities. In addition, it enables the <u>"balance law"</u> formulation, merging the finite volume and finite difference viewpoints.

The <u>cosistency</u> of the original Godunov scheme was an automatic (albeit somewhat implicit) consequence in the case of a piecewise-constant discretization. However, if higher order resolution is sought, this concept must be carefully studied. Here we offer a systematic study of consistency (and its order) for high order schemes (Section 4). The point to be emphasized is that it solely relates to the numerical flux and depends on the order of the underlying discretization space (Remark 4.16). Thus the two fundamental concepts of *numerical flux* and *consistency* are profoundly intertwined. Once they are established, we can proceed to state a generalization (to high order schemes) of the classical Lax-Wendroff convergence theorem.

7.1. **FUTURE DIRECTIONS.** We conclude by suggesting some research directions that represent natural extensions to the subject matter of this review.

• In the two-dimensional case, the Strang spatial splitting seems to be very effective when used on a regular grid. Beyond that, the focus should be turned to unstructured meshes:

(a) Establish suitable discretization spaces (piecewise constant, multilinear etc). In the piecewise-constant case, give a rigorous Godunov scheme.

(b) Determine a "Godunov-type", as well as higher order numerical fluxes, along each side of a mesh cell. In particular, such fluxes should account for transversal effects.

(c) Define a suitable consistency concept for such fluxes.

- Based on such fluxes, establish a generalization of the Lax-Wendroff convergence theorem. In particular, it is essential that the assumptions imposed in such theorems are amenable to verification in actual computations.
- Extend the treatment to systems of conservation laws with source terms (e.g. reacting flows or topography in shallow-water systems). The source terms play an important role in the derivation of local analytical solutions.

## References

- D. S.Balsara, J.Li and G. I.Montecinos, An efficient, second order accurate, universal generalized Riemann problem solver based on the HLLI Riemann solver, J. Comput. Phys., 375 (2018), 1238–1269.
- [2] J.B. Bell, P. Colella and J. A. Trangenstein, Higher order Godunov methods for general systems of hyperbolic conservation laws, J. Comput. Phys., 82 (1989), 362–397.
- [3] M. Ben-Artzi, The generalized Riemann problem for reactive flows, J.Comput.Phys. 81(1989), 70-101.
- [4] M. Ben-Artzi and A.Birman, Application of the "Generalized Riemann Problem" method to 1-D compressible flows with material interfaces, J.Comput.Phys. 65(1986), 170-178.
- [5] M. Ben-Artzi and A.Birman, Computation of reactive duct flows in external fields, J.Comput.Phys. 86(1990), 225–255.
- [6] M. Ben-Artzi and J. Falcovitz, A second order Godunov type scheme for compressible fluid dynamics, J. Comput. Phys., 55 (1984), 1–32.
- [7] M. Ben-Artzi and J.Falcovitz, An upwind second-order scheme for compressible duct flows, SIAM J.Sci.Stat.Comp. 7(1986), 744-768.

- [8] M. Ben-Artzi and J. Falcovitz, GRP-an analytic approach to high resolution upwind schemes for compressible fluid flow, in "Proceedings 9th Int. Conf. Numer. Methods Fluid Dynamics", (Eds. Soubbaramayer and Boujot), Springer Lecture Notes in Physics 218 (1985), 87–91.
- M. Ben-Artzi and J. Falcovitz, "Generalized Riemann Problems in Computational Fluid Dynamics", Cambridge Monographs on Applied and Computational Mathematics, Cambridge University Press, 2003.
- [10] M. Ben-Artzi, J.Falcovitz and U.Feldman, Remarks on high-resolution split schemes computations, SIAM J.Sci.Comp. 22(2000),1008–1015.
- [11] M. Ben-Artzi, J. Falcovitz and Ph. Le Floch, Hyperbolic conservation laws on the sphere. A geometry-compatible finite volume scheme, J. Comput. Phys., 228 (2009), 5650–5668.
- [12] M. Ben-Artzi, J. Falcovitz and J. Li, The convergence of the GRP scheme, Disc. Cont. Dynam. Sys., 23 (2009), 1-27.
- [13] M. Ben-Artzi and J. Li, Hyperbolic balance laws: Riemann invariants and the generalized Riemann problem, Numer. Math., 106 (2007), 369–425.
- [14] M. Ben-Artzi and J. Li, Consistency of finite volume approximations to nonlinear hyperbolic balance laws, Math. Comp., 90 (2021), 141–169.
- [15] M. Ben-Artzi and J. Li, Regularity of fluxes in nonlinear hyperbolic balance laws, Commun. Appl. Math. Comput. 5 (2023), 1289—1298.
- [16] M. Ben-Artzi, J. Li and G. Warnecke, A direct Eulerian GRP scheme for compressible fluid flows, J. Comput. Phys. 218 (2006), 19—43.
- [17] F. Bouchut, Ch. Bourdarias and B. Perthame, A MUSCL method satisfying all the numerical entropy inequalities, Math. Comp., 65 (1996), 1439–1461.
- [18] A. Bourgeade, Ph. Le Floch and P.A. Raviart, Approximate solution of the generalized Riemann problem and applications, in "Nonlinear Hyperbolic Problems", C.Carasso, P.A. Raviart and D. Serre (Eds.), Lecture Notes in Mathematics 1270 Springer-Verlag, 1986.
- [19] A. Bourgeade, Ph. Le Floch and P. A. Raviart, An asymptotic expansion for the solution of the generalized Riemann problem. Part 2: Application to the equations of gas dynamics, Ann. Inst. H. Poincaré, Nonlinear Analysis, 6 (1989), 437-–480.
- [20] Y. Brenier and S. Osher, Approximate Riemann solvers and numerical flux functions, SIAM J. Numer. Anal., 23 (1986), 259–273.
- [21] A. Bressan, The unique limit of the Glimm scheme, Arch. Rat. Mech. Anal., 130 (1995), 205–230.
- [22] C. Chainais-Hillairet, Second-order finite-volume schemes for a non-linear hyperbolic equation: Error estimate, Math. Meth. Appl. Sci., 23 (2000), 467–490.
- [23] G.-Q. Chen, M. Torres and W. Ziemer, Gauss-Green theorem for weakly differentiable vector fields, sets of finite perimeter, and balance laws, Comm. Pure Appl. Math., 62 (2009), 242– 304.
- [24] A.J. Chorin, Random choice solution of hyperbolic systems, J. Comput. Phys., 22 (1976), 517–533.
- [25] P. Colella and P. R. Woodward, The Piecewise Parabolic Method (PPM) for gas dynamical simulations, J. Comput. Phys., 54 (1984), 174–201.
- [26] F. Coquel and P.G.Le Floch, Convergence of finite difference schemes for scalar conservation laws in several space variables. General theory, SIAM J. Numer. Anal. **30** (1993), 675–700.
- [27] C. M. Dafermos, "Hyperbolic Conservation Laws in Continuum Physics, Fourth Edition", Grundlehren der Mathematischen Wissenschaften, vol. 325, Springer, 2016.
- [28] B. Despres, Lax theorem and finite volume schemes, Math. Comp., 73 (2004), 1203–1234.
- [29] X. Ding, G.-Q. Chen and P. Luo, Convergence of the fractional step Lax-Friedrichs scheme and Godunov scheme for isentropic gas dynamics, Comm. Math. Phys., 121 (1989), 63–84.
- [30] R. J. DiPerna, Convergence of approximate solutions to conservation laws, Arch. Rat. Mech. Anal., 82 (1983), 27–70.
- [31] Z. Du and J. Li, a Hermite WENO reconstruction for fourth order temporal accurate schemes based on the GRP solver for hyperbolic conservation laws, J. Comput. Phys., 355 (2018), 385–396.
- [32] Z. Du and J. Li, Accelerated piston problem and high order moving boundary tracking method for compressible fluid flows, SIAM J. Sci. Comput. **42** (2020), A1558-A1581.
- [33] V. Elling, A Lax-Wendroff type theorem for unstructured quasiuniform grids, Math. Comp., 76 (2007), 251–272.
- [34] L.C. Evans, "Partial Differential Equations", American Mathematical Society, 1998.

- [35] R. Eymard, T. Gallouët and R. Herbin, "Finite Volume Methods", in "Handbook of Numerical Analysis, Vol. VII", Eds. P.G. Ciarlet and J.-L. Lions, North-Holland, 2000, 713–1020.
- [36] J. Falcovitz and M. Ben-Artzi, Recent developments of the GRP method, Japan Soc. Mech.Eng. 38 (1995), 497–517.
- [37] H. Federer, "Geometric Measure Theory", Springer-Verlag, 1969.
- [38] T. Gallouët, R. Herbin and J.-C. Latché, On the weak consistency of finite volumes schemes for conservation laws on general meshes, SeMA Journal 76 (2019), 581–594.
- [39] E. Godlewski and P.-A. Raviart, "Hyperbolic Systems of Conservation Laws", Ellipses, 1991.
   [40] S. K. Godunov, Finite difference methods for numerical computations of discontinuous so-
- lutions of equations of fluid dynamics, Mat. Sb., 47 (1959), 271–295.
- [41] C. R. Goetz and M. Dumbser, A novel solver for the generalized Riemann problem based on a simplified LeFloch-Raviart expansion and a local space-time discontinuous Galerkin formulation, J. Sci. Comput., 69 (2016), 805–840.
- [42] J. B. Goodman and R. J. Leveque, A geometric approach to high resolution TVD schemes, SIAM J. Numer. Anal., 25 (1988), 268–284.
- [43] E.-E. Han, J.Li and H.Z. Tang, An adaptive GRP scheme for compressible fluid flows, J. Comput. Phys., 229 (2010), 1448-–1466.
- [44] A. Harten, High resolution schemes for hyperbolic conservation Laws, J. Comput. Phys., 135 (1997), 260-–278.
- [45] A. Harten, P. D. Lax and B. Van Leer, On upstream differencing and Godunov-type schemes for hyperbolic conservation laws, SIAM Review, 25 (1983), 35–61.
- [46] O. Igra, J. Falcovitz, H. Reichenbach and W. Heilig, Experimental and numerical study of the interaction between a planar shock wave and a square cavity, J. Fluid Mech., **313** (1996), 105–130.
- [47] V. P. Kolgan, Application of the principle of minimum values of the derivative to the construction of finite-difference schemes for calculating discontinuous solutions of gas dynamics, Scientific Notes of TsAGI, 3 (1972), 68–77.
- [48] D. Kröner, M.Rokyta and M. Wierse, A Lax-Wendroff type theorem for upwind finite volume schemes in 2-D, East-West J. Numer. Math., 4 (1996), 279–292.
- [49] S. N. Kružkov, First order quasilinear equations in several independent variables, Math. USSR Sbornik, 10(1970), 217–243.
- [50] A. Kurganov and E. Tadmor, New high-resolution central schemes for nonlinear conservation laws and convection-diffusion equations, J. Comput. Phys., 160 (2000), 241–282.
- [51] P. Lax, Shock waves and entropy, in "Contributions to nonlinear functional analysis" (Proc. Sympos. Math. Res. Center, Univ. Wisconsin ), 603–634, Academic Press, New York, 1971.
- [52] P. Lax and B. Wendroff, Systems of conservation laws, Comm. Pure Appl. Math., 13 (1960), 217–237.
- [53] P. G. Le Floch and T.Li, A Global asymptotic expansion for the solution to the generalized Riemann problem, Asymptotic Anal., 3 (1991), 321–340.
- [54] P. G. Le Floch and J.-G Liu, Generalized monotone schemes, discrete paths of extrema, and discrete entropy conditions, Math. Comput., 68 (1999), 1025–1055.
- [55] X. Lei and J. Li, A non-oscillatory energy-splitting method for the computation of compressible multi-fluid flows, Physics of Fluids 30 (2018); doi: 10.1063/1.5011093.
- [56] X. Lei and J. Li, Transversal effects of high order numerical schemes for compressible fluid flows, Appl. Math. Mech. -Engl. Ed., 40 (2019), 343–354.
- [57] J.Li, Two-stage fourth order: temporal-spatial coupling in computational fluid dynamics (CFD). Adv. Aerodyn. 1 (2019). https://doi.org/10.1186/s42774-019-0004-9
- [58] J. Li and G. Chen, The generalized Riemann problem method for the shallow water equations with bottom topography, Int. J. Numer. Methods Eng , **65** (2006) , 834–862.
- [59] J. Li and Z.Du, A two-stage fourth order time-accurate discretization for Lax-Wendroff type flow solvers, I. Hyperbolic conservation laws, SIAM J. Sci. Comput., 38 (2016), 3045–3069.
- [60] J. Li, T. Liu and Z. Sun, Implementation of the GRP scheme for computing radially symmetric compressible fluid flows, J. Comput. Phys., 228 (2009), 5867-5887.
- [61] J. Li, Q. Li and K. Xu, Comparison of the generalized Riemann solver and the gas-kinetic scheme for inviscid compressible flow simulations, J. Comput. Phys., 230 (2011), 5080–5099.
- [62] J. Li and Y. Wang, Thermodynamical effects and high resolution methods for compressible fluid flows, J. Comput. Phys., **343** (2017), 340–354.

- [63] J. Li and Q. Zhang, One-sided GRP solver and numerical boundary conditions for compressible fluid flows, J. Comput. Phys., 459 (2022), doi.org/10.1016/j.jcp.2022.111138.
- [64] T-T. Li and W-C. Yu, "Boundary value problems for quasilinear hyperbolic systems," Duke University Mathematics Series, V. 1985.
- [65] P.-L. Lions and P. E. Souganidis, Convergence of MUSCL and filtered schemes for scalar conservation laws and Hamilton-Jacobi equations, Numer. Math., 69 (1995), 441–470.
- [66] T.-P. Liu, Uniqueness of weak solutions of the Cauchy problem for general 2x2 conservation laws, J. Diff. Eqs., 20 (1976), 369–388.
- [67] X.-D. Liu, S. Osher and T. Chan, Weighted essentially non-oscillatory schemes, J. Comput. Phys., 115 (1994), 200–212.
- [68] M. Lukáčová-Medvid'ová and Y. Yuan, Convergence of a Generalized Riemann Problem scheme for the Burgers equation, Commun. Appl. Math. Comput. (2024). https://doi.org/10.1007/s42967-023-00338-x.
- [69] P.-H. Maire, A high-order cell-centered Lagrangian scheme for two-dimensional compressible fluid flows on unstructured meshes, J. Comput. Phys. 228 (2009), 2391–2425.
- [70] P.-H. Maire, A high-order cell-centered Lagrangian scheme for compressible fluid flows in two-dimensional cylindrical geometry, J. Comput. Phys., 228 (2009), 6882–6915.
- [71] I.S. Men'shov, Increasing the order of approximation of Godunov's scheme using the solution of the generalized Riemann problem, USSR Comput. Math. Math. Phys., 30 (1990),54–65.
- [72] K. W. Morton and D. F. Mayers, "Numerical Solution of Partial Differential Equations", Cambridge University Press, 2005.
- [73] S. J. Osher, Riemann solvers, the entropy condition, and difference approximations, SIAM J. Numer. Anal., 21 (1984), 217–235.
- [74] S. J. Osher, Convergence of generalized MUSCL schemes, SIAM J. Numer. Anal., 22 (1985), 947–961.
- [75] S. J. Osher and E. Tadmor, On the convergence of difference approximations to scalar conservation laws, Math. Comp., 50 (1988), 19–51.
- [76] J. Qian, J. Li and S. Wang, The generalized Riemann problems for compressible fluid flows: towards high order, J. Comput. Phys., 259 (2014), 358–389.
- [77] R. D. Richtmyer and K. W. Morton, "Difference Methods for Initial Value Problems", Wiley, 1967.
- [78] P.L.Roe, Approximate Riemann solvers, parameter vectors and difference schemes, J. Comput. Phys., 43 (1981), 357–372.
- [79] N. Seguin and J. Vovelle, Analysis and approximation of a scalar conservation law with a flux function with discontinuous coefficients, Math. Models Methods Appl. Sci. 13 (2003), 221–257.
- [80] C.-W. Shu, High order WENO and DG methods for time-dependent convection-dominated PDEs: A brief survey of several recent developments, J. Comput. Phys., **316** (2016), 598–613.
- [81] G. A. Sod, A Survey of several finite difference methods for systems of nonlinear hyperbolic conservation Laws, J. Comput. Phys., 27 (1978), 1–31.
- [82] E. Tadmor, A review of numerical methods for nonlinear partial differential equations, Bull. Amer. Math. Soc., 49 (2012), 507–554.
- [83] E.F. Toro, "Riemann Solvers and Numerical Methods for Fluid Dynamics: a practical introduction", Third edition, Springer-Verlag, 2009.
- [84] E. F. Toro, Derivative Riemann solvers for systems of conservation laws and ADER methods, J. Comput. Phys., 212 (2006), 150–165.
- [85] B. van Leer, Towards the ultimate conservative difference scheme.IV: A second order sequel to Godunov's method, J. Comput. Phys., 32(1979), 101–136.
- [86] B. van Leer, Upwind and high-resolution methods for compressible flow: From donor cell to residual- distribution schemes, Comm. Comp. Phys., 1 (2006), 192–206.
- [87] J. P. Vila, An analysis of a class of second-order accurate Godunov-type schemes, SIAM J. Numer. Anal., 26 (1989), 830–853.
- [88] Y. Wang and J. Li, Stiffened gas approximation and GRP resolution for compressible fluid flows of real materials, J. Sci. Comput. 95 (2023), https://doi.org/10.1007/s10915-023-02140-6
- [89] Y. Wang and S. Wang, Arbitrary high order discontinuous Galerkin schemes based on the GRP method for compressible Euler equations, J. Comput. Phys., 298 (2015), 113–124.

- [90] K. Wu, Z. Yang, H. Tang, A third-order accurate direct Eulerian GRP scheme for the Euler equations in gas dynamics, J. Comput. Phys. 264 (2014), 177–208.
- [91] Z. Yang, P. He and H. Tang, A direct Eulerian GRP scheme for relativistic hydrodynamics: One-dimensional case, J. Comput. Phys., 230 (2011), 7964-7987.
- [92] Z. Yang, H. Tang, A direct Eulerian GRP scheme for relativistic hydrodynamics: Twodimensional case, J. Comput. Phys., 231(2012), 2116-2139.
- [93] Q. Zhang and W. Sheng, The generalized Riemann problem scheme for a laminar two-phase flow model with two-velocities, J. Comput. Phys., 506(2024), 112929.

MATANIA BEN-ARTZI: INSTITUTE OF MATHEMATICS, THE HEBREW UNIVERSITY, JERUSALEM 91904, ISRAEL

Email address: mbartzi@math.huji.ac.il